Norden Beaches occurrent Beaches Store Busice Beaches Store Beaches Store Beaches



northern beaches council

Content

Key Terms And Definitions	4	
Executive Summary	6	
Section 1: Introduction	16	
Purpose of this Strategy	18	
Consultation	19	
Structure	20	
Section 2: Planning and Policy	22	
Greater Sydney's housing challenge	24	
Strategic policy context	26	
Local strategic planning	29	
Housing to achieve liveability and sustainability objectives	31	
Section 3: The Evidence	34	
LGA demographic and housing snapshot	36	
The housing continuum	42	
Housing demand	43	
Housing supply	51	
Housing capacity	54	
Demand and capacity gap	59	
Summary	60	

Section 4: The Right Locations For New Housing	62
Context	64
Options for medium to higher density renewal	71
Section 5: Specific Housing Types	76
Social and affordable housing	78
Affordable Rental Housing SEPP	80
Seniors accommodation	84
Low Rise Housing Diversity Code	88
Section 6: Sustainability Innovations	90
Innovative renewal	92
Renewable energy	93
Water efficiency	95
Sustainable mobility	98
Public domain improvements	100
Financial implications	101
Section 7: Future Planning	102
Allocation of housing type	104
Applying the Centres Renewal Framework	108
Future planning	126

Section 8: Strategy and Implementation	128
Overview	130
Priority 1 – Housing target	131
Priority 2 – Detailed planning for centres	132
Priority 3 – Social and affordable housing	133
Priority 4 – Precinct sustainability and housing	134
Priority 5 – Seniors housing	135
Potential statutory implementation pathways	136
Appendix 1 - Proposed housing diversity SEPP	138
Appendix 2 - Options considered for strategy	139
Appendix 3 - Summary of potential uplift by centre	144

Boarding houses	Buildings that provide multiple rooms for rent with generally shared communal facilities.			
Centres Renewal Framework	Approach taken in the Local Housing Strategy (LHS) for planning for new housing around key centres.			
Centre Investigation Areas	Areas within 800 metres of identified centres along the existing B-Line which will be investigated for future housing.			
СНР	Community Housing Provider – organisations which own and/or manage community housing dwellings.			
DPIE	NSW Department of Planning, Industry and Environment.			
DCP	Development control plan – sets out detailed rules for development within LGAs.			
Residential flat buildings	Includes dwellings in blocks of flats or apartments. These dwellings do not have their own private grounds and usually share a common entrance foyer or stairwell. This category also includes flats attached to houses such as granny flats, and houses converted into two or more flats. Also known as residential flat buildings.			
FSR	Floor space ratio – ratio of floor space to lot area permitted under planning controls.			
GSC	Greater Sydney Commission.			
GSRP	Greater Sydney Region Plan – Greater Sydney's strategic plan, developed by the GSC.			
LAHC	NSW Land and Housing Corporation.			
LEP	Local environmental plan – legislation which regulates land use and development within an LGA.			
LGA	Local government area.			
LHS	Local housing strategy – sets out a vision for housing in an LGA.			
LSPS	Local strategic planning statement – sets out a 20-year vision for land use in an LGA.			
Missing middle	Low-scale infill housing that can increase housing diversity in existing areas while being compatible with existing housing character.			

MRA	Metropolitan Rural Area – areas within the metropolitan boundary that are outside of the planned urban area. Generally refers to three or more dwellings (attached or detached) on one lot of land, including typologies such as terraces and townhouses.			
Multi-unit or multi-dwelling housing				
Other dwellings	Other types of dwellings that do not fit into standard separate, semi-detached or unit categorisations, such as caravans, cabins, houseboats or improvised homes.			
Private market housing	Housing owned or rented through the private market without government subsidy.			
RAI	Rental Affordability Index – a measure of the relative affordability of housing in different areas based on rents and incomes.			
SAH	Social and Affordable Housing – includes public housing, community housing and affordable rental housing, targeted to very low, low and moderate income households.			
Secondary dwellings	Self-contained dwellings established in conjunction with another dwelling on the same lot of land, which can be attached to or separate from the principal dwelling.			
Semi-detached dwellings	Includes semi-detached, row or terrace houses, townhouses and similar. These dwellings have their own private grounds and no other dwelling above or below them. They are either attached in some structural way to one or more dwellings or are separated from neighbouring dwellings by less than half a metre.			
Seniors living accommodation	Dwellings targeted to older people, including retirement villages, assisted living facilities and nursing home beds.			
Separate houses	es Houses separated from other dwellings by a space of at least half a metre. A separate house may have a flat attached to it, such as a granny flat or converted garage. Also referred to as detached housing.			
SEPP	State environmental planning policy – State-level planning instruments that set out rules for matters of State/regional environmental significance and can override LEPs.			

Introduction

This Northern Beaches Local Housing Strategy (LHS) looks at the mix of housing in the Northern Beaches Local Government Area (LGA) today, and at the kind of housing that will be needed in the future.

It considers trends in terms of population growth and change; household size and mix; lifestyle issues such as affordability, sustainability and building resilience; and housing diversity, including housing types such as boarding houses, seniors housing and social and affordable housing.

The LHS is shaped by the following vision:

As the Northern Beaches community grows and changes, residents will be able to the find the right housing that meets their needs, lifestyles and budgets.

This diverse mix of homes will be contained primarily within existing urban areas, with their design and construction respecting the area's heritage, environmental features and existing character.

New housing will be focused in and near centres where people can easily access public transport or walk or cycle to shops and services. This means that many parts of the LGA will only experience minimal change.

The mix of new housing will include well-designed flats, low rise dwellings, and social and affordable housing, including boarding housing. Seniors housing will be developed in areas that will allow older people to easily access transport and services.

This mix will mean that young and older residents can continue to live on the Northern Beaches in the community they know.

New housing will follow the principles of sustainable development, designed to reduce energy and water use and take advantage of natural elements such as breezes and heat. People will be less likely to use their cars and will enjoy living near new and enhanced open space areas. The LHS is informed by ongoing consultation, State and local strategic plans and policies, technical studies and analysis and an Issues and Opportunities Paper

The LHS will guide Council's planning for a diversity of housing in the right places. Overall, this means that many parts of the LGA will only experience very minimal change, ensuring that the LGA's character and environment will be retained.

Strategy and implementation

The LHS proposes a range of actions in five priority areas and identifies relevant NSW Government and other agencies who may need to assist to achieve the vision.

These five priority areas are:

- Priority 1: Housing targets - meet District Plan and 6-10 year housing target.
- Priority 2: Detailed planning for centres - establish sufficient capacity to accommodate housing demand around existing centres
- Priority 3: Social and affordable housing

 encourage the provision of affordable housing and plan for boarding house in appropriate and accessible locations
- Priority 4: Precinct sustainability and housing

 investigate and support
 sustainable housing precincts
- Priority 5: Planning for seniors housing - incentivise the provision of seniors housing in the right locations.

Priority 1: Housing targets

Meet District Plan and 6-10 year housing target.

The Northern Beaches will be home to a population of 288,431 people in 2036, an increase of 22,963 people from the 2016 Census. We need to plan for about 12,000 new dwellings by 2036.

We expect demand for separate houses with gardens will remain. However, suitable land for this type of housing is limited and environmental and other constraints make it difficult to plan for more of this type of dwelling. Instead, people looking for that type of housing will have the option to live in larger forms of attached or compact lowrise dwellings that offer people a small garden or courtyard, as well as residential flat buildings.

Most housing built in recent years was flats, units or apartments. We have also witnessed higher numbers of secondary dwellings (granny flats), boarding houses and seniors living developments. When we look at what can be achieved under current planning controls, and take into consideration the potential of the Frenchs Forest precinct (which is being planned by the NSW Government), we estimate there is capacity for a further 13,600 dwellings. However, we have also done a general analysis of what is financially feasible; this analysis finds that existing capacity is realistically around 11,700 dwellings.

This equates to a deficit of around 275 dwellings, which suggests a need to plan for additional capacity for housing in the LGA in the medium to longer term.

The LHS and its targets have been prepared based on projections and other information currently available. Post-COVID, projections may change and require updates to the LHS targets.

Priority 2: Detailed planning for centres

Establish sufficient capacity to accommodate housing demand around existing centres

New housing could be accommodated or distributed in different ways across the LGA. The LHS builds on the designation of strategic, local and village centres in Towards 2040, our local strategic planning statement. This allows us to plan for housing in a way that does not bring broadscale changes.

New housing will be focused on two different areas:

- Centre Investigation Areas
- Low-rise housing diversity areas

In the short term, low-rise housing diversity areas will be investigated within 400m of select centres for dual occupancies, seniors housing and boarding houses.

Centre investigation areas will be subject to detailed planning in the medium-long term to look at opportunities for all housing types within 800m of select centres along current and future B-line routes.

This means new housing will be located in places that are most convenient for residents and more sustainable for the whole community. NSW Government planning strategies, including the North District Plan, as well as Council's Towards 2040, look to locate medium and higher density housing in centres that have good accessibility to public transport.

Centre investigation areas

This strategy identifies Brookvale, Dee Why, Mona Vale, Manly Vale and Narrabeen along the existing B-Line as centre investigation areas in the medium term, and Forestville and Beacon Hill in the longer term, subject to a future B-Line route.

Dee Why already contains high density housing in the form of apartments, and only Brookvale and Frenchs Forest were identified in Towards 2040 as having longer term opportunities for higher density housing.

The focus of centre investigation areas therefore is to investigate opportunities for low-rise medium density housing around the centre, such as terraces and multi-dwelling housing.

Detailed planning for Brookvale will take into account its important role for local employment.

Low rise housing diversity areas

This strategy identifies Avalon, Newport, Warriewood, Belrose, Freshwater, Balgowlah and Manly as areas to support housing diversity in the form of dual occupancies, seniors housing and boarding houses.

This will create more diverse forms of housing across the LGA in locations with good access to shops and services.

Detailed planning for Manly will look to balance visitor accommodation with housing for permanent residents.

* Avalon Place Plan is currently under preparation and will have regard for the appropriateness of any density in this area"



Source: SGS, 2019.

Housing will be planned for using a Centres Renewal Framework for the centre investigation areas. This allows us to think about each area in terms of its unique collection and distribution of assets and attributes, within four precincts:

- Centre cores take advantage of immediate proximity to shops, services and transport and would typically include the most dense development.
- Mixed housing areas build on amenities and local assets with low impact medium density housing.
- Influence areas are where housing will be low to medium density and in keeping with local character.
- Excluded areas are constrained by existing built form, heritage or environmental aspects.

Additional housing capacity could be generated through the Centres Renewal Framework in each centre. This suggests additional capacity of between 1,540 and 2,930 dwellings is possible in the centre investigation areas, which would meet the gap in capacity.

However, we will undertake detailed planning for each centre. This planning will need to consider pathways for statutory implementation, including changes to both Council and NSW Government planning controls and policies, and the need for appropriate infrastructure to support the residents living in this new housing. Priority 9 of the Local Strategic Planning Statement seeks to ensure that provision of infrastructure accompanies growth in the housing and employment sectors in areas where urban renewal is occurring. The development of an LGA-wide land use infrastructure implementation plan will look to align all types of infrastructure planning with growth, based on housing, employment and social infrastructure studies and additional studies as required.

This will include detailed planning for each of the Centres Investigation Areas to determine the appropriate level of new development that can be accommodated.

This will consider potential impacts on local infrastructure, including transport, traffic, environmental sustainability and climate change, and social infrastructure such as schools. The need for new infrastructure will also be informed by other Council strategies, such as the Social Infrastructure Study and Open Space and Recreation Strategy.

The Centres Investigation Areas will be considered individually to reflect their unique character and circumstances. This means that what is planned in Mona Vale will not be the same typologies or scale of development centres like Dee Why or Brookvale.

Centres renewal framework approach



Priority 3: Social and affordable housing

Encourage the provision of affordable housing and plan for boarding houses in appropriate and accessible locations

Housing affordability is an issue for many residents on the Northern Beaches. As well as advocating for action from the NSW and Australian governments, we are committed to a comprehensive social and affordable housing plan so that more people can afford to live in their Northern Beaches community.

Demand for boarding houses, as an alternative affordable housing option should be met in locations around centres, with incentives to encourage provision on suitable larger sites.

We can start to tackle affordability and provide options for more people by planning for the right **diversity of housing** that will allow more people to be able to afford to live in the Northern Beaches. This includes:

- low density housing, such as single and secondary dwellings or low to medium density housing, such as attached dual occupancies
- medium density housing, including those allowed by the Low Rise Housing Diversity Code
- higher density residential flat buildings
- social and affordable housing
- seniors housing
- boarding houses.

We've identified a current shortfall of around 8,100 social and affordable housing dwellings; this will increase by a further 1,880 dwellings by 2036. For the first time ever, Council will adopt **a target to plan for 1,880 new social and affordable housing dwellings by 2036** and will continue to advocate for other levels of government to address the significant current shortfall.

We also want to address the identified demand for an additional 102 single boarding house rooms.

We will seek exemption from several State Environmental Planning Policies (SEPPs) which provide for different housing e.g. boarding houses and seniors housing. Some State-level planning policies are not suited to the Northern Beaches, and both Council and the community have expressed concerns with State-level planning policies, including the potential impacts of ad-hoc development.

We propose seeking an exemption from the Low Rise Housing Diversity Code in the Northern Beaches, given the demand for medium density housing can be met through the strategies and implementation of the LHS through LEPs. These housing forms will still be permitted, but instead targeted to areas around centres with good accessibility and amenity.

Priority 4: Precinct sustainability and housing

Investigate and support sustainable housing precincts

Towards 2040 commits Council to several sustainability, liveability and other targets that require innovative and progressive approaches to planning for housing and centres, particularly at a precinct level. We will therefore use **new development as a means to leverage greater sustainability and liveability outcomes.**

Sophies Gardens, North Manly



Priority 5: Planning for seniors housing

Incentivise the provision of seniors housing in the right locations

We will encourage seniors housing in accessible locations, enable the industry trend of the 'continuum of care' approach on larger sites, and provide incentives for seniors housing that meets the needs of the community.

There is an identified demand for an additional 1,716 self-contained retirement village units ('standard' housing generally targeted to over 55s), 502 assisted living units (with some support facilities) and 765 nursing home beds (offering end of life care) by 2036. A Northern Beaches-specific approach will:

- allow for the redevelopment and upgrade of older and larger seniors housing sites to increase the supply of all-inclusive seniors housing including residential care facilities
- incentivise the provision of seniors housing in accessible locations in Centre investigation Areas, within multi-level developments
- limit developments in inaccessible locations that will not meet the needs of older people.

Warriewood



Low Density

		Single dwelling	Two dwellings (secondary dwellings)	Dual occupancy (attached, and detached on corner lots only)	Multi-dwelling housing and terraces
	E4 Environmental Living/R5 Large Lot Residential				
	Rest of R2 zones (not including Centres investigation areas and not within 400m of centres)				
	Low rise housing diversity areas (400m radius)				
Cantre investigation areas	Throughout Centre Instigation Area (approx. 800m radius) (Influence area)				
	Mixed housing - 400-800m (to be defined through detailed planning work)		1 		
Can	Higher density and mixed use core - up to 800m (to be defined through detailed planning work)		 		

Single dwelling

Two dwellings (secondary dwellings)

Dual occupancy (detached/side by side)



Source: SGS, 2020.

High Density Dual occupancy **Boarding houses Residential flat** (one above Manor house Seniors housing buildings other)

Dual occs

(one above other) Manor house

Residential flat buildings

Section 1 Introduction



The NSW Department of Planning, Industry and Environment (DPIE) framework for the development and implementation of local housing strategies (LHSs) in NSW² requires all NSW councils to prepare a LHS.

The Northern Beaches LHS sets a vision for future housing in the LGA and how we will plan for emerging housing needs arising from the area's changing demographics and pressures on housing affordability. It demonstrates how the housing directions within the North District Plan will be delivered locally. The LHS means that new housing will be provided in the places where people can have greater access to transport and services, and that more people will be able to live in homes that suit their needs, lifestyles and budgets. Most areas in the LGA will only be subject to minimal changes.

The LHS is informed by a range of data sources, including consultation, State and local strategic plans and policies, detailed background analysis and development of an evidence base. It is also informed by the **Issues and Opportunities** Paper (exhibited as part of Towards 2040, Council's local strategic planning statement). This paper identified key housing issues, established a housing vision and explored spatial options and land use planning approaches for the LGA.

Figure 1

local housing policy development framework



Source: SGS, 2019, based on DPIE Local Housing Strategy Guideline.

² A Guideline and Template identified a desirable process and structure for Local Housing Strategies, see <u>https://www.planning.nsw.gov.au/-/media/Files/DPE/</u> <u>Guidelines/Local-Housing-Strategy-Guideline-and-Template.pdf</u>

Consultation

Through the consultation undertaken, the community told us they:

- support greater housing diversity and affordable housing to enable people from diverse backgrounds, young people and key workers to live in the community, as well as the need for affordable seniors housing
- support a range of mechanisms for affordable housing including seeking support from the NSW Government
- have mixed views (though generally supportive) on seeking exemptions from State policies for seniors living, boarding houses and medium density housing to enable a local approach to planning for these development types, with some concerns around how this could impact affordability and housing diversity
- prefer a limit on high rise development, and instead would prefer Council to focus on subdivisions to provide low to medium density housing and more housing diversity around accessible centres
- are resistant to continued population and housing growth without accompanying infrastructure, particularly public transport
- want the LGA to retain its low density and family housing character.

Benefits of strategic planning

A long-term, strategic approach helps to:

- plan for new and different forms of housing not traditionally available in Northern Beaches LGA as community demographics change (including young people forming new households, and ageing of the population)
- address the lack of affordable housing, including to support key workers who will also be needed as the population ages
- address liveability issues (such as minimising traffic impacts and maximising access to open space), support the shops and services provided in local centres and avoid ad-hoc housing development that limits the planning for and provision of supporting infrastructure
- protect the environmental and lifestyle quality of the Northern Beaches by rethinking how centres are planned and how new housing is provided.

This LHS will allow us to respond to changing conditions and ensure sufficient and suitable housing to support the community through their different life stages. The LHS is structured as follows:

- Section 2 outlines the existing planning and policy context concerning housing on the Northern Beaches.
- Section 3 sets out the evidence base, including analysis of future housing demand and the capacity of current planning controls to accommodate the demand.
- Section 4 outlines the right locations for new housing.
- Section 5 identifies proposed approaches to add to the stock of social and affordable housing, plan for seniors accommodation in the right locations and direct low to medium density development to appropriate locations.

- Section 6 discusses how sustainability and liveability objectives will require innovative approaches to planning for new housing in and around centres.
- Section 7 summarises the approach to the planning and development of new housing across the LGA.
- Section 8 sets out the vision for housing and an implementation plan.





Manly Beach/North Steyne Active Travel Path



Section 2: Planning and Policy



In 2016, Greater Sydney was home to around 4.7 million people. The Department of Planning, Infrastructure and Environment (DPIE) forecasts that by 2036, the city will be home to 6.6 million people, and increase of more than 1.9 million people or 42 per cent, in 20 years. As a comparison, between 1996 and 2016, Greater Sydney grew by around 1.1 million people.

Planning for this growth while maintaining the quality of life offered in Sydney's suburbs requires us to consider how to best accommodate more people in established areas as well as at Sydney's fringes. Each part of Greater Sydney will see an increase in the number of people and dwellings, including the Northern Beaches LGA.

The Northern Beaches LGA covers more than 25,000 hectares, with a range of different environments. It stretches along the coastline and into the hinterland, from Manly in the south to Palm Beach in the north, and extends west into Ku-Ring-Gai Chase National Park.

According to the Australian Bureau of Statistics (ABS), the estimated residential population of the Northern Beaches LGA in 2016 was 265,468. DPIE's population projections released in 2019 forecast a population of 288,431 people by 2036.

This is an additional 22,963 people – an 8.7 per cent increase or 0.4 per cent annual growth rate. The DPIE projections indicate the growing population will require an additional 14,803 dwellings between 2016 and 2036 (although these projections were formulated prior to the outbreak of COVID-19). Historically most population growth in Greater Sydney has been accommodated through:

- the development and conversion of rural and agricultural land at Sydney's fringes
- the redevelopment of former industrial sites for high density apartment complexes
- the construction of higher density apartment housing around existing mixed use centres and public transport such as Dee Why
- the consolidation of established residential neighbourhoods, including policies that allow development of dual occupancy dwellings as well as forms such as villa housing suitable for seniors, as is the case in Warriewood.

All these types of development continue. However, planning for the Northern Beaches requires us to consider what kinds of housing will best meet the needs of the local community and be most suitable in the context of demographic shifts, climate change and environmental challenges.

Sydney's urban boundary cannot continue to expand without limit, and dwellings built in existing suburbs near public transport are often better located than similar dwellings built on the urban fringe, which are often far away from public transport, job and services.

Impacts of COVID-19

The COVID-19 pandemic has had and will continue to have many impacts. Restrictions on travel and migration within Australia and internationally are likely to affect levels of population growth, particularly in the short term. Given the uncertainty, it is not possible to accurately predict the impact that COVID-19 associated restrictions will have on rates of population growth, demand for particular housing types, or the need for affordable housing. This reiterates the need for effective, flexible planning for the future of housing on the Northern Beaches and Sydney. The LHS may need to be reviewed to take account of new or updated population projections.



Source: SGS, 2019.

Greater Sydney Region Plan and North District Plan

The Greater Sydney Region Plan envisages a metropolis of three cities and a 30-minute city, where residents enjoy quicker and easier access to a range of jobs, housing types and activities.

The Region Plan splits Sydney into five districts, with the Northern Beaches in part of the North District. The Region Plan and North District Plan include objectives to increase overall housing supply and ensure that housing becomes more diverse and affordable.

The North District Plan includes a 20-year housing target of an additional 92,000 dwellings by 2036 across the North District. The Northern Beaches has a five year housing supply target to 2021 of 3,400 additional dwellings. This represents 13 per cent of the target for the North District.

The North District Plan stresses the need to locate new housing in areas with local infrastructure. It identifies locational criteria for urban renewal and local infill development. Towards 2040 and this LHS reflect these criteria

Councils must establish 6-10 year housing targets when developing their LHS, while also informing potential future changes to planning controls in relevant local environmental plans (LEPs). We have also identified a 20-year projection of housing demand to assist with long-term planning. This will be regularly reviewed.

Figure 2

Proportion of five-year housing target by Iga in the north district



Source: SGS, 2019.

State environmental planning policies

Existing state environmental planning policies (SEPPs) that relate to housing include:

- State Environmental Planning Policy (Housing for Seniors or People with a Disability) 2004
- State Environmental Planning Policy (Affordable Rental Housing) 2009
- State Environmental Planning Policy (Exempt and Complying Development Code) 2008.

The Housing for Seniors and People with a Disability SEPP establishes requirements for where this type of housing should be located (near public transport, of a minimum size, and not in bushfire prone land). Sites that meet these criteria are dispersed across Northern Beaches LGA.

The Affordable Rental Housing SEPP allows for boarding house and secondary dwelling developments in the R1 General Residential, R3 Medium Density Residential, R4 High Density Residential, B1 Neighbourhood Centre, B2 Local Centre and B4 Mixed Use zones, and in the R2 Low Density Residential zone on sites that accessible to public transport. Many sites in the Northern Beaches meet these criteria. The Exempt and Complying Development Code allows for a streamlined assessment process for development that complies with specific standards, such as those that have a limited environmental impact. The SEPP applies across all zones except for developments under the Affordable Rental Housing SEPP.

Proposed NSW Housing Diversity Sepp

DPIE exhibited the Housing Diversity SEPP -Explanation of Intended Effect in 2020. This sets out a proposal to prepare a new Housing Diversity SEPP that would consolidate and update the Affordable Rental Housing 2009, Housing for Seniors and People with a Disability SEPP and State Environmental Planning Policy No 70–Affordable Housing (Revised Schemes) (SEPP 70). Appendix 1 summarises the proposed changes

Low Rise Housing Diversity Code³

The Low Rise Housing Diversity Code allows for a fast-tracked development process for complying approvals for some types of dwellings in the R1 General Residential, R2 Low Density Residential, R3 Medium Density Residential and RU5 Village zones. This includes manor houses, one and two storey dual occupancies, and terraces.

The application of the Code was temporarily deferred in the Northern Beaches LGA, but came into effect as of July 2020. Some amendments have been made to Manly LEP 2013 in response to the application of the Code in the LGA, including the introduction of a minimum site area per dwelling and minimum allotment sizes for multi-dwelling housing and manor houses in the R2 Low Density Residential Zone.⁴

Exemptions To State Policies

Council will seek exemptions from the application of SEPPs for:

- Seniors housing under the Housing for Seniors or People with a Disability SEPP
- Boarding houses under the Affordable Rental Housing SEPP
- Low Rise Housing Diversity Code.

Instead, we will use the strategies and directions in this LHS to meet demand for these types of housing in areas around centres with good accessibility and amenity.

Frenchs Forest precinct

The Frenchs Forest precinct at the new Northern Beaches Hospital is identified by the NSW Government as a centre that will become a new destination and focal point for the community. In 2017, Northern Beaches Council prepared the Hospital Precinct Structure Plan, which will guide the planning and development of the precinct. The Frenchs Forest precinct will accommodate around 4,360 dwellings and is factored into the LHS.

Other planning work

Other relevant Council strategic planning work underway includes the development of the **Brookvale Structure Plan**, and the potential for future residential development in **Ingleside**, with a need to consider bushfire risk and other environmental constraints.

³ Previously known as the Low Rise Medium Density Housing Code.

⁴ See <u>www.northernbeaches.nsw.gov.au/planning-and-development</u> /planning-future/planning-growth/low-rise-housing-diversity-code

Local planning for housing issues

Our existing policies and plans identify several issues for the LGA, including need for:

- more affordable housing social and affordable housing stock is at historically low levels, with wait times for public housing over five years, and median property prices and rents have increased significantly
- better transport connectivity the LGA has limited public transport, meaning any new housing will require investment in appropriate accompanying infrastructure
- sustainable growth that maintains the character of the LGA - the natural environment and amenity is both a key strength yet also a constraint to future housing development.
- greater housing diversity this will be required to cater to different demographic groups and household types, including older people.

SHAPE 2028 Northern Beaches Community Strategic Plan and the MOVE – Northern Beaches transport strategy focus on sustainable growth, concentrating more intensive development around existing centres, and linking future growth to improvements in transport.

Local Strategic Planning Statement

These themes are also reflected in Towards 2040, which guides future land use in the LGA, reflecting local values and aspirations, and building on the 10-year vision set out in SHAPE 2028.

Towards 2040 includes:

- a 20-year vision
- planning priorities to guide local land use planning
- principles that underpin planning priorities and actions
- actions Council will take to help achieve the priorities
- measures of success and an implementation program to determine whether priorities have been achieved.

Towards 2040 Vision

"In 2040, the Northern Beaches has a stunning coastal and bushland environment, an enriched and contemporary coastal character and better connections to the North District and the rest of Greater Sydney.

The natural environment is healthy and protected and highly valued by residents and visitors alike. There is a range of housing to accommodate the whole community and we continue to pursue design excellence and sustainability outcomes in built forms. It offers a thriving local economy and a sustainable mix of employment and industrial lands and vibrant and enlivened centres.

The healthy and active community can easily access artistic, creative, sporting and recreational opportunities and the services and facilities that support their health and wellbeing." Towards 2040 contains 30 priorities under four themes: sustainability, infrastructure and collaboration, liveability and productivity. Two priorities relate to housing:

- **Priority 15** Housing supply, choice and affordability in the right locations
- **Priority 16** Access to quality social housing and affordable housing.

A series of principles underpin these priorities.

Towards 2040 also includes other priorities, principles, actions and measures that will influence how we plan for new housing in the LGA. Another key priority is Priority 13 - Strong Engagement and cooperation with Aboriginal communities. Council is committed to undertake this, including engagement with the Metropolitan Aboriginal Land Council (MLALC) as a large landowner in the Northern Beaches LGA. ⁵

Many of these require development to respect existing character, contribute to sustainability and make the Northern Beaches a better place to live.

Table 1:

Housing priorities and principles in the Northern Beaches LSPS

Priority	Principles				
Priority 15 Housing supply,	 Locate new housing in strategic and local centres and within reasonable walking distance (800m) of high-frequency public transport. 				
choice and affordability in the right locations	Provide greater housing diversity and affordable housing options.				
	 Limit development where there are unacceptable risks from natural and urban hazards, or impact on tree canopy 				
	• Use existing urban land more efficiently to protect the natural environment.				
	 Safeguard employment lands from non-compatible uses, particularly residential and mixed-use development. 				
	 Ensure new buildings are high amenity and do not unreasonably impact on neighbouring properties and the public domain. 				
	Encourage adaptable and universal design				
	 Sequence growth with provision of public transport, open space and other infrastructure in strategic and local centres. 				
	• Ensure new housing is designed to complement local character, heritage and the environment				
	• Contribute to a public benefit and better urban design outcomes.				
	 Ensure new residential development is located within 400m of open space and high density areas within 200m of open space. 				
	• Ensure development is low-carbon with high-efficiency in energy, water and waste.				
Priority 16 Access to quality social housing	 Build more high-quality and diverse social housing and affordable rental housing in areas close to services and facilities and with access to high-frequency public transport. 				
and affordable housing	Protect and retain existing affordable rental housing.				
	 Seek a minimum 10% affordable rental housing for all planning proposals for upzoning, urban renewal or greenfield development, with higher rates where financially feasible. 				
	 Ensure the available supply of affordable rental housing is shared equitably among vulnerable residents. 				

Source: Northern Beaches Council, 2019.

⁵ The MLALC has been in long-term discussions with both Council and DPIE in relation to a Belrose site (known as Lizard Rock), and has requested formal inclusion of this site in SEPP (Aboriginal Land) 2019 via Ministerial approval of a Development Delivery Plan (DDP). The proposal is under review by DPIE.

Housing to achieve liveability and sustainability objectives

The Region Plan and North District Plan include directions for the scope of an LHS, the framework for the quantitative analysis of future housing demand and supply, and locational criteria in planning for housing.

Towards 2040 identifies housing, liveability and sustainability priorities that will require an integrated approach to planning and development including innovations in housing renewal.

For example, the target to increase the proportion of dwellings with access to open space will not be achieved if new housing is not provided near parks or if the land development process is not utilised to create additional open space. Similarly, increasing the proportion of dwellings within 30 minutes of a strategic centre will be best achieved by providing additional housing

189 McPherson St, Warriewood

in areas with good transport access. Achieving targets around the diversity of new housing and providing more affordable housing will also require the development of new, mixed housing stock.

Achieving targets around energy and water usage, waste recycling and the people-centric outcomes related to community will also rely on new approaches in how housing is designed and utilised.

Accommodating new housing growth to meet the future needs of the population is an opportunity to make a real difference in achieving sustainability and liveability outcomes.

As much of the LGA's built form is established and unlikely to change significantly under current controls, creating opportunities for new approaches and innovations in building design and planning becomes crucial.



Final December 2021

32 Figure 3

Directions and priorities in the LSPS

Direction	Priorities			
A Contraction of the second se	Priority 1 Healthy and valued coast and waterways	Priority 2 Protected and enhanced bushland and biodiversity	Priority 3 Protected scenic and cultural landscapes	Priority 4 Protected Metropolitan Rural Area
√ ŧÕj ∕ Efficiency	Priority 7 A low-carbon community, with high energy, water and waste efficiency			
Resilience	Priority 8 Adapted to the impacts of natural and urban hazards and climate change			
Infrastructure and collaboration	Priority 9 Infrastructure delivered with employment and housing growth			
	Priority 10 World-class education facilities, including a university	Priority 11 Community facilities and services that meet changing community needs	Priority 12 An inclusive, healthy, safe and socially connected community	Priority 13 Strong engagement and cooperation with Aboriginal communities
Housing	Priority 15 Housing supply, choice and affordability in the right locations	Priority 16 Access to quality social and affordable housing		
Great places	Priority 17 Centres and neighbourhoods designed to reflect local character, lifestyle and demographic changes	Priority 18 Protected, conserved and celebrated heritage		
Connectivity	Priority 19 Frequent and efficient regional public transport connections	Priority 20 Sustainable local transport networks	Priority 21 Redesigned road space and facilities to match changing community needs	
لت (الله) Jobs and skills	Priority 22 Jobs that match the skills and needs of the community	Priority 23 Frenchs Forest as a sustainable health and education precinct	Priority 24 Brookvale as an employment and innovation centre	Priority 25 Dee Why as a thriving cosmopolitan centre by the sea
Themes: • Sustainabil	ity • Infrastructure and co	llaboration • Liveability	 Productivity 	

Priority 5 Greener urban environments Priority 6

High quality open space for recreation

Priority 14

A community enriched through the arts and connected through creativity

Priority 26

Manly as Sydney's premier seaside destination

Priority 27

Mona Vale as the contemporary, urban heart of the north Priority 28 Safeguarded employment lands **Priority 29** A thriving, sustainable tourism economy **Priority 30** A diverse night-time economy

Section 3: The Evidence



LGA demographic and housing snapshot

Further detail on the demographics and other characteristics of the LGA is available in the Issues and Opportunities Paper.

Figure 4:

Demographic overview - Average five year growth rate, 2006-2016

Population

- The Northern Beaches LGA saw growth of 27,000 people between 2006 and 2016, at a rate of around six per cent every five years.
- Growth has been of a similar rate to the North District, but lower than across Greater Sydney.
- Most growth has been concentrated around Brookvale and Dee Why.
- Population densities overall have increased between Census years.



Figure 5:

Demographic overview - Age Profile, 2016

Age Profile

- The strongest growth between 2006 and 2016 has been in the retiree (people aged 65+ years), mature adult (45-65 years) and youth populations (5-20 years).
- Proportions of children (0-5 years), young adults (20-30 years) and adults (30-45 years) dropped in this period.
- The age profile varies geographically Manly, Brookvale and Dee Why have lower proportions of retirees and higher proportions of young and working age adults compared to the rest of the LGA.


Figure 6:

Demographic overview - Household Types, 2016

Household Types

- Couples with children are the most common household type, accounting for 35 per cent of households.
- The highest rate of growth in households between 2011 and 2016 was also in couple households with children.
- Northern Beaches has a slightly higher proportion of couples with children, single parent and couple only households compared to the North District and Greater Sydney.



Figure 7:

Demographic overview - Number of bedrooms spare, 2016

Households and dwelling types

- Couple family households tend to occupy detached dwellings (similar to North District and Greater Sydney) while less than a third live in attached dwellings or apartments
- Higher density dwellings typically house couples without children and single person households

 however, the proportion of families with children living in apartments has increased.
- In larger dwellings there are relatively high proportions of houses with spare bedrooms, similar to trends in the North District and Greater Sydney. Around 55 per cent of four-bedroom dwellings have two or more bedrooms spare.



Dwelling types

Separate houses remain the most common form of housing in the LGA, making up over 50 per cent of dwellings in 2016. Semi-detached dwellings account for around 10 per cent of dwellings, and apartments around 35 per cent. Compared to the North District and Greater Sydney, the LGA has a slightly higher proportion of detached houses. However, between the 2011 and 2016 Censuses, semi-detached dwellings and apartments increased as a proportion of total dwellings.

Table 2:

Dwelling Types, Northern Beaches LGA, 2016

LGA	Separate house	Semi-detached	Residential flat buildings	Other
Total dwellings	56,537	9,548	34,541	438
Proportion of total	56%	9%	34%	0%

Source: ABS 2016 Census.

Predominant dwelling types vary between suburbs across the LGA. Areas further inland, such as Terrey Hills and Frenchs Forest, have much higher shares of separate houses. Coastal locations and areas around the major centres, like Manly and Dee Why, have much higher proportions of semi-detached and apartment dwellings.



Source: SGS, 2019, based on ABS 2016 Census.

Social and affordable housing

The need for social and affordable housing remains an issue for the Northern Beaches given generally high property prices and rents. Median dwelling prices in the Northern Beaches LGA have grown substantially in recent years. As of September 2020, the median rent in the LGA was \$650 per week, compared to \$495 across Sydney. The median dwelling price at this time was also over \$1.5 million, compared to only \$690,000 for the rest of Sydney. The Northern Beaches LGA has historically also had lower proportions of social housing than the rest of the metropolitan area. ⁶

The limited affordability of the Northern Beaches is reflected in the results of the Rental Affordability Index (RAI). The RAI measures rental affordability using the 30/40 rule in relation to household incomes, where households in the lowest 40 per cent of incomes paying more than 30 per cent of their income on rent are considered in housing stress.

Rents in the Northern Beaches LGA have been rated as Unaffordable to Extremely Unaffordable for a typical rental household. This compares unfavourably to many other parts of Greater Sydney. The lack of affordable housing is pronounced for households on lower incomes, with affordability for couples earning the minimum wage and single person households on benefits rated as Severely Unaffordable to Extremely Unaffordable.

⁶ Department of Communities and Justice Rent and Sales Statistics, March 2021.



Housing comes in many forms and can be planned for through a range of mechanisms and approaches. The housing continuum (see Figure 8) illustrates the broad spectrum of approaches in terms of the public, community and private sectors. Most housing is provided by the private sector for homeowners and private renters on the right end of the continuum. Moving left along the continuum are the forms of housing for lower income households where subsidies are typically provided. **Social housing** is the traditional State-owned housing for low income renters. At the extreme left end of the spectrum is crisis and emergency accommodation types, for homeless people or victims of domestic violence for example, which require the highest level of government subsidy.

Figure 8:

Housing Supply Contiuum



Source: SGS, 2020.

In the NSW context and in this report the types of housing that are discussed are:

- private market housing for owning or renting (this may include housing for lower income tenants whose rent may be subsidised through Commonwealth Rent Assistance)
- boarding houses (both traditional and new forms) which would typically be priced at the lower income end of the market housing spectrum though with some tenants receiving rent subsidies (and some operators getting concessions as well)
- seniors living dwellings as permitted by the Housing for Seniors or People with a Disability SEPP, again within the market spectrum though including some lower income occupiers receiving rent subsidies
- social and affordable housing types as defined by SEPP 70 for very low to low income households, which are likely to be managed by the community housing sector and by definition are rented and subsidised to a greater or lesser extent depending on household income.

Drivers of housing demand

Strong demand for housing in the Northern Beaches comes particularly from buyers seeking to downsize. Properties with low maintenance requirements are popular. Vacancy rates overall remain low across the LGA.

While house prices are high overall compared to many areas of Greater Sydney, the Northern Beaches market can be seen as 'two-speed'. There is often strong demand from downsizers at the higher or luxury end of the market, with drop offs observed at the lower end, possibly influenced by bank lending restrictions. COVID-19 may further flatten demand for lower priced housing, particularly apartments for rent and purchase.

A key driver of demand in the Northern Beaches remains the attractiveness of its lifestyle and proximity to the natural environment. Broader factors such as transport and other infrastructure investment can influence housing demand. While the lack of heavy rail has been a constraint, current and proposed future investments, such as Warringah Road and Wakehurst Parkway around the Northern Beaches Hospital, Beaches Link, and better public transport connections to Chatswood and Macquarie Park will contribute to housing demand.

Impacts of COVID-19

Changes to migration patterns, population growth and the housing market due to COVID-19 are expected, but it is difficult to accurately assess how significant these trends will be over the short, medium and longer term.

Real impacts in the property market can already be observed, with many areas in Australia seeing reduced demand for housing and consequently lower rent and house prices.

While there could be reduced demand for housing from lower population growth, there is still a need for greater housing choice, and for place-based planning to ensure that Council can meet the changing needs of the community over time.

Future housing demand

The housing demand analysis has been updated from data in the Issues and Opportunities Paper to reflect updated population projections.

Change in population and households

As the population increases, the share of household family types across the Northern Beaches will remain relatively consistent other than an increase in the proportion of single person households. Couple families with children will remain the most common type of household to 2036.

Projected housing demand

Projections of the future demand for housing in the LGA are derived from population projections that use the 2016 Census as a base. There is estimated to be demand for an additional **14,803 dwellings** between 2016 and 2036 in the LGA, or the need for an average of approximately 740 dwellings per year to 2036.

44 Table 3:

Projected northern beaches housing demand to 2036

Dwelling type	2016	2021	2026	2031	2036	Change 2016-36	Average annual growth rate
Separate house	60,236	60,801	61,537	62,555	63,539	3,303	0.27%
Semi-detached house	9,641	10,268	11,054	11,941	12,872	3,231	1.46%
Flat, unit or apartment	35,308	36,863	38,940	41,398	44,016	8,707	1.11%
Other dwelling	1,272	799	781	807	833	-439	-2.09%
Total private dwellings	106,458	108,730	112,312	116,701	121,261	14,803	0.65%

Source: SGS, 2020. Based on DPIE 2019 population projections.⁷

Dwelling development between 2016 and July 2020 has contributed towards meeting the modelled housing demand. This is summarised in the table below, with the result that the remaining housing demand for the LGA from 2020 to 2036 is 12,018 dwellings.

Additional secondary dwellings are included in this table, while other kinds of dwelling are not, because secondary dwellings are often built in low density residential zones on properties which may not otherwise have capacity for development, and as such secondary dwellings are not included in the additional capacity analysis presented in Section 3.5. The large number of separate houses in the Northern Beaches means that there will be limited capacity constraints on the delivery of additional secondary dwellings to 2036. Most future demand is expected to be for residential flat buildings, followed by semi-detached houses and separate houses. Apartments and semi-detached dwellings are also projected to increase as a proportion of dwellings, while the proportion of detached houses is expected to decrease to around 52 per cent by 2036.

While demand for detached houses will continue, this may not be possible given environmental and other constraints. A share of the demand may be channelled into larger forms of attached dwellings that offer similar characteristics such as ground level entrances and private open space areas. These are more likely to be provided as infill development.

Projected population growth represents a smaller proportional increase over the period to 2036 than what is expected for both the wider North District and Greater Sydney. The population of the Northern Beaches is projected to grow by around nine per cent, compared to 23 per cent for the North District and 42 per cent for Greater Sydney. However, unlike other LGAs in Greater Sydney, the Northern Beaches does not have major transport infrastructure (such as a train line) to support substantial growth in an efficient and sustainable way.

⁷ Note: the DPIE population projections were released before the COVID-19 pandemic.

The LHS may need to be reviewed to reflect revised projections.

Table 4:

Northern Beaches remaining housing demand calculation

Modelled demand 2016-2036	
Total housing demand	14,803
Private dwellings completions 2016 – July 2020	
Additional private dwellings	2,120
Private dwellings demolished [1]	736
Increase in private dwellings (Additional dwellings – dwellings demolished)	1,384
Other dwelling completions 2016-July 2020	
Boarding house rooms	268
Seniors living dwellings	247
Secondary dwellings	709 (25% of which is 177)
Total contribution to housing demand (with 25% of secondary dwellings counted [2])	+692
Dwelling demand July 2020 - 2036	
Potential additional secondary dwellings July 2020 – July 2036 (assuming the same yearly rate of development as between 2016 - July 2020, and that 25% of secondary dwellings contribute to meeting housing demand)	+709
Total remaining demand (Housing demand 2016-2036 - Increase in private dwellings – total contribution from other completions - potential additional secondary dwellings)	12,018

This estimate is calculated from Building Approvals Australia, Australian Bureau of Statistics, (September 2020), adjusting for a proportion of approvals not being acted upon

 ^[2] This is similar to the average percentage of secondary dwellings with rental bonds lodged in the SSROC area as reported in research by the City Futures Research Centre in Affordable Housing SEPP and Affordable Housing in Central and Southern Sydney (May 2020)

Demand for social and affordable housing

The social and affordable housing demand analysis has been updated since the Issues and Opportunities Paper to reflect updated population projections.

Social housing includes subsidised public and community housing, with eligibility defined by the NSW Government. It means people with low incomes can live independently or find housing that suits their needs. Eligible households are placed on the NSW Housing Register.

Affordable housing is different to social housing. It can be made available to a broader range of households, including for key workers in health and services who live on low to moderate incomes. Affordable housing may be managed more like a private rental property (though still with eligibility criteria) mostly by notfor-profit community housing providers.

The modelling that informs this LHS models social and affordable housing collectively, using a combination of Census data and the NSW Affordable Housing Guidelines. In 2017, the 1,201 social housing tenancies in the Northern Beaches LGA (not including community housing) were mainly one bedroom or studio units. Management has recently transferred to community housing providers. This represents around 1.2 per cent of housing stock, compared to the rate of around 3.7 per cent in Greater Sydney as of the 2016 Census.

In 2016, there was estimated to be demand for more than 9,000 social and affordable housing dwellings in the LGA from households experiencing either moderate or severe rental stress or who are homeless. The modest supply of around 1,000 social and affordable housing dwellings means there is a shortfall of 8,100 dwellings, or around eight per cent of total stock). There may also be additional current unmet demand from households living elsewhere who are unable to afford to live in the Northern Beaches.

Demand will increase by around 1,800 households by 2036.

Table 5:

Projected Northern Beaches demand for social and affordable housing to 2036

Dwelling type	2016	2021	2026	2031	2036	Change
Couple family with children	1,617	1,638	1,625	1,622	1,614	-3
Couple family with no children	1,870	1,889	1,992	2,114	2,247	377
Families (sub-total)	3,487	3,527	3,617	3,736	3,862	375
Group household	640	623	632	648	671	31
Single person household	3,737	3,966	4,278	4,627	4,998	1,261
One parent family	1,277	1,345	1,399	1,452	1,494	217
Total	9,141	9,461	9,926	10,463	11,025	1,884

Demand will be driven mainly by single person households, consistent State-wide trends that reflect an older population and more complex household compositions. Future demand will also shift with changing affordability conditions in terms of wage growth and house prices.

Table 6:

Breakdown of projected Northern Beaches demand for social housing and affordable housing in 2036

Household type	2016	2021	2026	2031	2036	Change
Households in social housing	1,694	1,782	1,902	2,034	2,174	480
Homeless households	605	636	681	734	788	183
Households in housing stress	6,842	7,043	7,343	7,695	8,063	1,221
Total	9,141	9,461	9,926	10,463	11,025	1,884

Source: SGS, 2020. *Note: this assumes that social housing dwellings will increase in line with population growth.

Demand for boarding houses

Boarding house developments include traditional boarding houses and so-called 'new generation' boarding houses (defined as 'co-living' housing in the proposed Housing Diversity SEPP). New generation boarding houses tend to have fewer shared facilities compared to traditional boarding houses in dwellings that are more self-contained. Under the right controls, boarding houses provide an alternative affordable housing option.⁸

Based on the boarding house share of total demand, and applying this to population projections, an additional 343 people will need to be accommodated in boarding houses by 2036, equivalent to a demand for 343 single boarding house rooms or 172 double boarding house rooms. This projected demand is based on the projected population growth between 2016 and 2036, yet an additional 241 boarding house rooms have entered operations since 2016.⁹ Together, these rooms could accommodate between 241 and 399 lodgers based on room sizes and specified capacity.

If 241 lodgers are accommodated by these boarding houses, only a further 102 single rooms will be needed, equivalent to five new developments (based on the average development of 20 rooms,¹⁰ noting however that the number of rooms is limited to 12 under the current SEPP). Northern Beaches projected growth in boarding house lodgers, 2016-2036

	2016 (from ABS Census)	2036 (projection)	Difference (projected-existing)
Number of boarding house lodgers	102	445	343

Source: SGS, 2020

Factors influencing demand

Future demand will be influenced by price and boarding house type. Recent development applications largely reflect the new generation boarding houses that tend to have limited shared facilities, and are only slightly cheaper than renting in the private rental market.

While these new generation boarding houses can meet the needs of key workers and other professionals working in the Northern Beaches, they do not meet the affordability needs of people in low income groups. As of April 2020, the average rent for a new generation boarding house in the LGA was around \$436 per week – which is unaffordable for many people.¹¹ Although the projections forecast modest demand for boarding houses to 2036, previous research indicates there are gaps in the market for affordable boarding house options in Sydney.¹² We need to plan for consistent supply to provide more affordable housing in the right locations, including in boarding house forms, by community housing providers. Actual or revealed demand may ultimately be higher or lower than our current estimates, particularly given the proposed changes in the Housing Diversity SEPP may shift market fundamentals for this type of housing.

⁸ The proposed NSW Housing Diversity SEPP included an affordability requirement that all boarding houses need to be affordable and managed by a registered not-for-profit community housing provider.

⁹ Based on Northern Beaches Council occupation certificate data.

¹⁰ Based on existing boarding house development in the Northern Beaches.

¹¹ Based on 12 recent listings sourced from homely.com.au and rent.com.au.

¹² See UNSW 2018, 'State Environmental Planning Policy (Affordable Rental Housing) 2009 and affordable housing in Central and Southern Sydney,' City Futures Research Centre, <u>https://cityfutures.be.unsw.edu.au/research/projects/affordable-housing-sepp-and-southern-sydney/</u>

Pittwater Park and Barrenjoey Road



Demand for seniors accommodation

Demand will increase for all types of seniors housing as an increasing proportion of the population is over 55. Seniors typically live in their own homes, retirement villages or nursing homes. Each provides a different level of care.

Healthy seniors who require limited support may choose to live in their own homes and receive occasional home care services.

Those with some functional impairment or who wish to live in a community setting may choose retirement villages. These offer different levels of care: assisted living units provide additional care for those who need daily assistance, while self-contained living offers a lower level of assistance suited to more independent residents.

Seniors who have severe impairments and need higher levels of care may need to live in nursing homes, or residential aged care, where personal and nursing care are provided daily. We expect there will be a high level of demand for retirement villages and nursing homes by 2036.

Table 8:

Northern Beaches projected supply gap for seniors housing

	2016 (from ABS Census)	2036 (projection)	Difference (existing-projected)
Retirement villages – self-contained independent living units	4,196	6,256	2,060
Retirement villages – assisted living units	993	1,495	502
Nursing homes (beds)	1,500	2,265	765

Source: SGS, 2020.

An additional 128 self-contained independent living units have been developed since 2016¹³ helping to address some of these gaps for self-contained independent living units. This leaves additional demand of 1,716 self-contained units to 2036. Assuming a median of 81 units per development¹⁴ the demand is equivalent to around 21 new developments. There will also be additional demand for assisted living and nursing home units, though at a lower level than for self-contained units.

¹³ Based on Northern Beaches Council occupation certificate data.

¹⁴ Based on existing seniors housing development in the Northern Beaches.

Residential zoning

The predominant residential land use zones in the Northern Beaches LGA are R2 Low Density Residential, R3 medium Density Residential. The E4 Environmental Living zone also accommodates residential dwellings. Residential dwellings are also permitted in some business zones, such as B1 Neighbourhood Centre, B2 Local Centre, and B4 Mixed Use. However, the LGA's current zoning scheme differs between the former LGAs of Manly, Warringah, and Pittwater, and their respective LEPs:

- Manly LEP 2013
- Warringah LEP 2011
- Warringah LEP 2000 (Deferred Matter areas)
- Pittwater LEP 2014.

McDonald Street, Freshwater





Source: SGS, 2020.

Drivers of housing supply

Consistent with trends revealed by the Census, since 1992, multi-unit housing has accounted for 59 per cent of dwelling approvals and around 55 per cent of approvals since 2013. Dwelling completions since 2013 have reflected this trend even further, with around 80 per percent of new dwellings being multi-unit. There has also been a recent increase in the number of secondary dwellings in the LGA, with around 709 developed between 2016 and 2020. As well as this, 268 boarding house rooms, and 247 seniors living units were completed in the same period.

Northern Beaches Local Housing Strategy

Figure 9:

Dwelling Approvals, Northern Beaches LGA, Nov 2013 - Dec 2019



Figure 10:

Dwelling Completions, Northern Beaches LGA, Nov 2013 - Dec 2019



Source: DPIE Land and Housing Monitor, 2020.

Source: DPIE Land and Housing Monitor, 2020.

Future land release areas

Opportunities to develop new land release areas in Northern Beaches LGA are limited, with environmental constraints likely to restrict the volume of possible residential development across the LGA.

Warriewood is the key remaining land release area, though much of it is already developed. Ingleside was previously identified as a location for a development; however, expectations have been scaled back due to environmental hazards and constraints.

In relation to Ingleside, there is potential for a maximum of approximately 1000 dwellings. This is scheduled for potential public consultation to occur in the first half of 2021 and the final dwelling numbers will be subject to the outcome of that process. *The capacity analysis below has been updated since the Issues and Opportunities Paper to reflect new data.

Notional housing capacity

Housing capacity estimates the quantum of housing that could be accommodated in an area, based on what would be allowed under existing planning controls and recent housing supply trends. Estimates can only be indicative rather than absolute.

Figure 11: Capacity Assessment method



Source: SGS, 2020.

The theoretical capacity assessment suggests that if all available areas are fully developed, the LGA could accommodate around 8,250 additional dwellings in addition to the maximum of 5,360 dwellings identified for Frenchs Forest and Ingleside, which is not factored into current planning controls.

Combined, **this suggests an overall capacity for around 13,600 dwellings.** Much of the existing capacity is for multi-unit housing, residential flat buildings and shop-top housing.

Table 9:

Notional housing capacity by dwelling type

Highest permissible dwelling type	Existing number of developable properties	Existing number of dwellings	Maximum number of dwellings	Raw capacity
Dual occupancies	409	395	818	423
Dwelling houses	41	17	136	119
Multi-dwelling housing and residential flat buildings	599	899	5,700	4,801
Shop-top housing	254	207	3,045	2,838
Density-controlled lots*	52	52	127	75
Total	1,355	1,570	9,826	8,256
Frenchs Forest Planned Precinct				4,360
Ingleside precinct				1,000
Total including Frenchs Forest				13,616

Source: SGS, 2020. *Density-controlled lots are located in an area in Manly that has a dwelling density control.

Location of notional capacity

The identified existing capacity is generally concentrated around centres, including Dee Why, Brookvale and Mona Vale. As would be expected there is a comparatively limited amount of capacity in rural and environmentally sensitive locations.

Map 5

Location of notional capacity under current controls (mesh block level*)



Much of the capacity for new housing is in the B2 Local Centre, B4 Mixed Use and R3 Medium Density Residential zones.

Table 10:

Notional dwelling capacity by zone, Northern Beaches LGA

Zone	Existing number of developable properties	Existing number of dwellings	Maximum number of dwellings	Raw capacity
B1 Neighbourhood Centre	60	55	409	354
B2 Local Centre	175	180	1,882	1,702
B4 Mixed Use	55	89	1,718	1,629
B5 Business Development*	27	4	530	526
E3 Environmental Management	1	2	3	1
E4 Environmental Living	6	4	10	6
R1 General Residential	34	36	86	50
R2 Low Density Residential	341	321	737	417
R3 Medium Density Residential	549	787	4,235	3,448
R5 Large Lot Residential	48	48	99	51
RU2 Rural Landscape	58	45	116	72
RU4 Primary Production Small Lots	1	-	1	1
Total	1,355	1,570	9,826	8,256
Frenchs Forest Planned Precinct				4,360
Ingleside precinct				1,000
Total including Frenchs Forest				13,616

57

Source: SGS, 2020. *Housing is an additional permitted use in B5 zone in Brookvale.

Feasible capacity

A form of sensitivity testing of the theoretical capacity considered the financial feasibility of new development in the LGA. This factors in construction, land value and other costs and compares these to potential sale prices for different types of housing. This high level model reduces the available housing capacity in the LGA from 8,250 dwellings to around 6,400 dwellings including feasible and marginally feasible sites (excluding Frenchs Forest and Ingleside).

Table 11:

Indicative feasible capacity by dwelling type, Northern Beaches LGA

Development type	Feasible development	Marginally feasible development	Unfeasible
Multi-dwelling housing and residential flat buildings	3,570	375	857
Shop-top housing	1,967	221	650
Dual occupancies	106	58	259
Dwelling houses	45	16	58
Density-controlled lots*	9	18	48
Total	5,695	688	1,872

Source: SGS, 2020. *Density-controlled lots are located in an area in Manly that has a dwelling density control.

Most shop-top housing and apartment development is considered as feasible, while infill development further away from centres is generally less feasible. Site-specific environmental and other constraints will also limit the potential for development in some areas. There is a notional gap of around 3,700 dwellings between the projected housing demand from 2020 and 2036 and capacity under current controls. This gap is addressed when the capacity identified in Frenchs Forest Planned Precinct and Ingleside is included. Potential secondary dwellings in low density residential areas are not included in the capacity analysis, assuming that they continue to be built at the same rate as they have since 2016, and that 25% of them take up housing demand.

However, when factoring in the potential feasibility of development there is a deficit in capacity of around 275 dwellings even when including Frenchs Forest and Ingleside.

Table 12:

Northern Beaches LGA housing capacity gap in 2036

	Estimated number of dwellings (excluding Frenchs Forest)	Including Frenchs Forest Planned Precinct
Notional capacity under controls	8,256	13,616
Projected additional demand to 2036 (from est. 2020 levels)	12,018	12,018
Difference (capacity – demand)	-3,762	1,598
Feasible development capacity (feasible and marginally feasible)	6,383	11,743
Projected additional demand to 2036 (from est. 2020 levels)	12,018	12,018
Difference (capacity – demand)	-5,635	-275

Source: SGS, 2020.

Considering the gap between notional capacity and demand in terms of the feasible capacity illustrates the development likely to occur under current controls (recognising that feasibility will change over time). If the feasible capacity is insufficient, changes to planning controls will be required to accommodate future growth. However, this is not considered necessary in Northern Beaches LGA in the very short term.

Some of the identified capacity gap may also be addressed through current planning work such as the Brookvale Structure Plan.

Population growth and change in demographics

Projected population growth will drive demand for new and different forms of housing. For example, the increase in the number of single person households drives a need for smaller households, as does an older population. Combined with affordability issues we expect to see greater demand for smaller and lower maintenance housing, such as residential flat buildings. While demand for larger dwellings remains, only a very small share of this can be met due to environmental constraints and a lack of suitable land. Demand will partly be met by attached, secondary or compact dwellings in established areas that offer features like courtyards or ground-floor access.

Housing affordability

As rents and prices rise, housing is severely or extremely unaffordable for many tenants, in particular. There is a shortfall or around 8,100 social and affordable rental dwellings; this will increase by an additional 1,880 dwellings over the 20-year period between 2016 and 2036.

Housing capacity

Noting the deficit in capacity of around 275 dwellings, the current planning framework is not sufficiently addressing the need for more diverse housing choices, including medium density housing. Increased diversity will address people's changing needs and provide more affordable smaller housing to help address affordability.

These issues also require us to plan for alternative housing forms such as boarding houses or different types of seniors accommodation near locations with shops and services. Recent residential development around Dee Why



Section 4: The Right Locations For New Housing



The structure plan in Towards 2040 and housing opportunities map sets out a spatial framework for the LGA and identifies key centres. The housing opportunities map within the Housing Priority identified opportunities for housing diversity. Towards 2040 is the strategic reference point for this LHS and our approach to housing renewal in different centres across the LGA. It means that the majority of the LGA will see little change, with new development focused on centres.

Under Towards 2040, strategic centres (Dee Why, Brookvale, Mona Vale, Manly and Frenchs Forest) service higher order needs, serve as hubs for employment, and typically have higher density forms of housing. Local centres (Avalon, Newport, Warriewood, Narrabeen, Freshwater, Manly Vale, Balgowlah, Forestville and Belrose, as identified in the North District Plan), are local hubs for employment services. Of these 9 Local Centres, the LSPS identified opportunities for housing diversity along the existing B-Line bus route (Manly Vale, Brookvale, Dee Why, Narrabeen, Warriewood and Mona Vale) and Manly. Beacon Hill and Forestville were identified for housing diversity subject to a B-Line extension and a new east-west B-Line bus route.

⁵⁷ Campbell parade, Manly Vale



In addition to the centres identified in the structure plan, we propose to investigate housing diversity around the local centres of Avalon, Belrose, Freshwater and Balgowlah, as these have amenities and assets that could support modest additional development in the future.

Analysis draws from a number of spatial options, including:

- concentrated urban renewal in and around strategic centres
- dispersed urban renewal in and around all centres
- transport-centric renewal
- 'missing middle' housing diversity.

Appendix 2 details these options further

Centre investigation areas

The proposed approach for new housing applies the principles of Towards 2040 and aims to build in long term capacity for growth around centres with good transport, while respecting each centre's scale and character of and increasing housing diversity and affordability.

This would see medium density development concentrated in strategic centres and selected town centres (current and future B-Line routes), while these and other smaller centres on transport corridors would be a focus for greater housing diversity. Most streets and suburbs in the LGA will remain very much the same as today. Priorities for renewal in the short to medium term will be centres on the existing B-Line: Brookvale, Dee Why, Mona Vale, Manly Vale and Narrabeen. Brookvale, Dee Why and Mona Vale are already strategic centres and Manly Vale and Narrabeen offer opportunities to build on their existing small-centre characteristics. small-centre characteristics (though noting that the outcomes of Council's coastal and resilience studies may result in Narrabeen having limited opportunities). Other centres on the B-Line, such as Warriewood, are only identified for low-rise housing diversity opportunities, due to local environmental and other constraints. ¹⁵

With structure and employment planning studies underway, Brookvale could be the initial focus, followed by Dee Why and Mona Vale. Planning for Manly Vale and Narrabeen could follow in the medium-term.

Frenchs Forest is not considered in detail in this LHS, as its development is subject to NSW Government planning at this time.

In the longer term, if a second B-Line to Chatswood goes ahead, centres such as Forestville and Beacon Hill in this corridor could also be a focus.

Renewal in each centre will reflect how the centre is used in the LGA – for example, renewal in strategic centres will be different to that in smaller centres. Additional development will avoid areas with absolute development risks (see Section 7.2.) Plans for renewal will also reflect the existing character of each centre - meaning that what is suitable for Dee Why will be different to what is suitable in Mona Vale.

¹⁵ In the case of Warriewood, the location of the B-Line stop and the layout of the wider centre does not lend itself to renewal. There are also issues with flooding and nearby land uses that would make concentrated renewal more challenging.

66 Map 6:

Towards 2040 structure plan



Map 6 Legend

Urban area	 Forest village 	📰 Employment hub
Ingleside Growth Area	 Coast walk 	O Existing housing capacity
National park	Bondi to Manly walk	ldentified future housing supply
Major public parkland/	Ferry routes	
bushland	 Ferry stop 	Regional community facilities
Metropolitan Rural Area (MRA)	 Public transport and active 	
Future MRA investigation area	travel connection	Education
Strategic centre	Future B-line link to Newport	🚥 Sports and recreation
Health and education precinct	Key transport corridor	Art/culture/community
Local centre	Future transport corridor	Primary medical facility
Coastal village	 Beaches Link Tunnel investigation area 	 Supporting medical facility

Source: Northern Beaches Council, 2019.

67



Low-rise housing diversity areas

Other centres may also be suitable locations for some low to medium density housing to develop greater diversity of housing across the LGA, and particularly in locations that are close to shops, services, clubs and community facilities.

Local centres suited to low to medium density housing such as dual occupancies, seniors accommodation and boarding houses include the strategic, town and village centres that are not on the existing B-Line or not proposed for significant housing renewal. These include:

- Avalon ¹⁶
- Newport ¹⁷
- Warriewood
- Belrose
- Freshwater
- Balgowlah
- Manly.

For these centres, Council may apply a road network/pedestrian shed to refine the distance from the town centre and a range of other criteria to assess those properties that are suitable for change.

We recognise Manly and Balgowlah already permit small lot sizes and a broad range of housing types within the 400m radius, including dual occupancies, terraces, residential flat buildings, seniors development and boarding houses.

A Place Plan is being prepared for Manly that will look to balance visitor accommodation with housing for permanent residents.

The preferred approach will be to include low to medium density typologies in residential areas within 400 metres of these centres, excluding areas with significant development constraints (see Section 7.2).

- ¹⁷ Newport Village Commercial Centre Masterplan was adopted in 2007.
- ¹⁸ Forestville and Beacon Hill may be investigated for renewal subject to the establishment of the second B-line route.

¹⁶ A Place Plan is being prepared for Avalon which will guide any development in this village.



Development in the centres identified for medium to high density housing should aim for:

- walkable and public transport accessible neighbourhoods
- diverse housing character precincts
- higher density housing in mixed use centres to minimise the impact on established areas
- medium and high density development near amenities and services
- low-scale infill for more diverse housing in existing areas which is compatible with existing housing character ('missing middle')
- heritage and environmental areas to be protected.

Missing Middle Housing

The 'missing middle' refers to low-scale infill housing that can increase housing diversity in existing areas while being compatible with existing housing character. This can include housing types such as dual occupancies, manor houses, and multi-dwelling forms like terraces.

Traditional concentrated urban renewal approach

The traditional urban design approach to planning for renewal around centres focuses higher density development in areas within 400 metres from the central node, such as a town centre or a public transport stop, with medium to high density development between 400 metres and 800 metres from the node. Examples of this type of development include Dee Why which has seen high density apartment development, particularly within 400 metres of the core node.

This approach concentrates new housing into locations where people can access transport or services that are also further from existing areas. However, it tends to favour apartment development, which is not necessarily suited to all locations, doesn't provide a mix of housing and doesn't allow for more moderate infill development that could better reflect the character of existing areas and contribute to their reinvigoration.

Figure 12:

traditional concentrated urban renewal approach



Linear urban renewal approach

A linear approach to urban renewal focuses on transport corridors, with more intense development around key nodes, and higher density development along the corridor in between nodes. This approach has been taken in the Canterbury Road and Parramatta Road corridors in Sydney's Inner West.

Figure 13: Linear urban renewal approach



Source: SGS, 2020.

This approach can take advantage of existing or new transport services on road corridors, and it may provide cheaper forms of entry-level housing.

However, it can affect the issues that make a place a nice place to live, in terms of noise, pollution or amenity, and in terms of health outcomes. Action 81 of the North District Plan specifically recommends avoiding these 'urban hazards'. This pattern of renewal also does not necessarily enhance public transport access for people living between nodes, and not within walking distance of a stop.
Dispersed development

The dispersed approach to renewal implies an ad-hoc approach, with individual developments on sites dispersed across an LGA, including through spot rezonings. It may occur if an industrial area is rezoned for housing yet not close to transport. This type of development has been a feature of many locations in Greater Sydney through the recent housing boom.



Dispersed development



Source: SGS, 2020.

While this approach facilitates a supply pipeline and requires little additional planning, its haphazard nature means that new housing won't necessarily reflect the qualities and character of the area nor achieve a liveability and sustainability dividend from growth. Instead, it creates pressure for supplementary or new transport or social infrastructure and services in locations or ways that might not be optimal or efficient. This approach does not meet the criteria of the North District Plan and priorities of Towards 2040.

Centres Renewal Framework

Given this approach, we propose a **Centres Renewal Framework** that identifies different housing precincts in developable (unconstrained) areas around a centre or public transport stop, within an 800 metre, or a 10-15 minute, walk. These areas are defined as **Centre Investigation Areas**.

Figure 15:

Centres renewal framework

Defining the investigation area boundary:

- Areas within 800 metres of centres
- Exclude areas with heritage, topography or environmental constraints



Precincts

1. Centre core

- Highly accessible and in town centre areas less than 800m to public transport.
- Areas that are logical extensions of a centre where people can easily walk to the centre without facing physical barriers.
- Includes larger sites suitable for redevelopment, or areas with potential for site amalgamations to host additional development.
- Suited to mixed use developments (with ground floor retail) and higher density housing such as apartments (such as within Brookvale and Frenchs Forest).

2. Mixed housing

- Up to 800m from centres and public transport.
- Good amenity, representing opportunity areas closest to assets such as parks and shops.
- Suited to larger lot sizes, typically over 600 sqm, ready for redevelopment without site amalgamations.
- Suitable for housing such as townhouses and small scale apartments of 2-4 storeys.

3. Influence areas

- Located within within 800m walk of centres,
- Suitable for housing that match the character of existing detached housing areas, such as dual occupancies, terraces, semi-detached dwellings or manor homes.
- Ideally creating Torrens (not strata) titled properties.

4. Excluded areas

 Heritage conservation areas, sites with high environmental hazards and risks, and areas zoned for E4 Environmental Living.

74

Source: SGS, 2020.

This approach allows for a range of housing types, with a lower density overall, that can be designed in keeping with neighbourhood character yet build on local assets and features and account for constraints. It enables a broader approach to how a centre might function in terms of aspects such as walking and cycling access or integrated planning for open space and other neighbourhood amenities.

Table 13:

Comparison of potential approaches to housing renewal

Approach	Pros	Cons
Traditional concentrated urban renewal	 Locates more housing in most accessible locations 	 Favours apartment development only Limited flexibility to reflect constraints and retain character
Linear urban renewal	 Takes advantage of provision of transport infrastructure 	 Provides limited amenity/accessibility unless located adjacent to a centre May concentrate housing in unhealthy traffic corridors Can impact built form character and quality of streetscape
Dispersed development	 Requires limited additional strategic planning/largely driven by market forces and easy supply opportunities 	 Haphazard development which may not reflect existing housing character Limited ability to plan for additional pressures placed on infrastructure Likely to generate community angst
Centres Renewal Framework	 Allows for a variety of housing in accessible locations Flexible application that considers existing features and character; excludes constrained areas Allows for comprehensive planning, integrating assets Allows for new housing to respect local conditions 	 Wider area of potential disruption around centre, but still focussed Requires more detailed planning analysis

Source: SGS, 2020.

Section 5 Specific Housing Types



While we will set a target and strategy to facilitate more social and affordable housing the NSW and Australian governments hold the overwhelming responsibility for providing shelter for households in need of assistance.

Our contribution will be to at least meet the additional demand for social and affordable housing dwellings between 2016 and 2036 – that is, a minimum target for 1,880 dwellings. The gap between this and the total demand for social and affordable housing in 2036 (including the existing unmet demand) should be addressed by other levels of government.

Comprehensive approach to delivery

There are a range of approaches and mechanisms that could be considered to help meet the social and affordable housing target.

Affordable housing contribution scheme

Under Towards 2040, Council endorsed including a target for a minimum of 10 per cent affordable rental housing in new planning proposals, consistent with Council's existing Affordable Housing Policy and its Affordable Housing Contribution Scheme under SEPP 70.

This alone will not achieve the target amount of additional social and affordable housing, without higher density forms of housing; other mechanisms will be needed to complement the affordable housing contributions scheme.

Other mechanisms

Voluntary planning agreements: Affordable housing contributions could be negotiated where planning proposals or other development concessions provide floor space in addition to that provided in planning controls negotiated through voluntary planning agreements (i.e. separate from the SEPP 70 Contribution)

Inclusionary zoning: Like the long-term approach in Ultimo/Pyrmont where a small percentage of all new floor space is dedicated to affordable housing (it can be paid, and typically is, as an equivalent cash contribution), this approach acknowledges that everyone benefits when lower income households, including workers who are essential to community functioning, have affordable housing. Requirements to provide car parking spaces are a type of 'inclusionary zone'.

An inclusionary zone for affordable housing could apply at, for example, three to five per cent of all new residential floor space, and perhaps at a lesser rate on non-residential floor space. Cash in lieu contributions could be collected by a nominated community housing provider to invest in affordable housing just as CityWest Housing does in Ultimo/Pyrmont.

The NSW Government would need to approve such an approach.

Underutilised or surplus Council-owned or

community-owned sites: Surplus or underutilised land "(not including existing open or green space)" could be developed for affordable housing or a mix of private, social and affordable housing. Council land such as that used for car parking could be redeveloped with parking retained in any new development or commercial or retail sites could be redeveloped with shop-top housing. Community land may include surplus church-owned sites.

Direct cash contributions: Council could make an annual cash contribution to a community housing provider or may dedicate a share of revenue from its commercial properties, rates or other revenue. Port Phillip Council in Melbourne has seed funded a community housing provider. This option may better meet the aim of providing affordable housing in better locations and would recognise that Council's property base represents an asset similar to its financial resources. Joint ventures: These might see Council working with the NSW Land and Housing Corporation (LAHC) to redevelop existing social housing on the Northern Beaches by assisting with approvals, waiving developer contributions or providing a cash or land contribution.

These various mechanisms are charted below. While we can estimate the number of dwellings that a contributions scheme might fund, this approach would likely fall short the minimum target without a combination of the above approaches. This will require a comprehensive strategy to identify the potential and scope of these other mechanisms.

Figure 16:

Comprehensive approach to meeting the social and affordable housing target



Source: SGS, 2020

The Affordable Rental Housing SEPP¹⁹ guides the development of new infill affordable housing, secondary dwellings, boarding houses, supportive accommodation, LAHC development, group homes, and residential flat buildings provided by social housing providers, public authorities and joint ventures.

The Affordable Rental Housing SEPP defines a boarding house as "a building that is wholly or partly let in lodgings, provides lodgers with a principal place of residence for three months or more, may have shared facilities, and has rooms that accommodate one or more lodgers". Boarding houses do not include backpackers' accommodation, group homes, serviced apartments, seniors housing, or hotel or motel accommodation.

Both traditional and new generation boarding houses are covered under the Affordable Rental Housing SEPP.

Locational controls

The Affordable Rental Housing SEPP allows boarding houses in the following zones with some additional locational and other requirements.

Table 14:

Locational controls for boarding houses under the affordable rental housing sepp

Permissible zones	Additional controls
R1 General Residential	-
R2 Low Density Residential	Only permissible in accessible locations (within 800m walking distance of a railway station/ferry wharf; within 400m walking distance of a light rail; within 400m walking distance of a regularly used bus stop) No more than 12 rooms per development
R3 Medium Density Residential	-
R4 High Density Residential	-
B1 Neighbourhood Centre	Ground floor will not be used for residential purposes
B2 Local Centre	Ground floor will not be used for residential purposes
B4 Mixed Use	Ground floor will not be used for residential purposes

Note: Boarding houses are currently permitted within the deferred lands area as identified within Warringah LEP 2000 Source: State Environmental Planning Policy (Affordable Rental Housing) 2009.

¹⁹ The Affordable Rental Housing SEPP as it applies to boarding houses may be rolled into the proposed Housing Diversity SEPP.

Several changes to the Affordable Rental Housing SEPP concerning boarding houses are summarised below.

Table 15:

Recent changes to provisions for boarding houses

Year	Item	Description
2011 May	Character of local area	Developments need to be deemed by the consent authority as compatible with the character of the area
2018 June	Car parking requirement	Car parking requirement was increased from 0.2 spaces per room in accessible areas, and 0.4 spaces in non-accessible areas to 0.5 spaces per room in all areas for boarding houses not provided by a community housing provider
2019 Feb	Boarding houses in Zone R2 Low Density Residential	Boarding houses in R2 Low Density Residential zone cannot exceed 12 rooms per site to manage the bulk and scale of boarding house developments in the low-density zone

Source: State Environmental Planning Policy (Affordable Rental Housing) 2009.

Application in Northern Beaches LGA

The Affordable Rental Housing SEPP and its amendments have not resulted in more affordable housing options for the Northern Beaches.

To increase the provision of affordable rental and social housing in the Northern Beaches, we will seek an exemption from some of the requirements of the Affordable Rental Housing SEPP and introduce local planning provisions to facilitate boarding house development. These provisions are as follows:

- Boarding houses developments are to be permissible in R2, B1, B2 zones only when within 400 metres of local centres on sites greater than 1,000 square metres with a maximum of 12 rooms and developed and maintained in perpetuity by a community housing provider.
- Boarding house developments are only permitted in locations within Centre Investigation Areas around centres on the B-Line (likely R3, R4²⁰ and B4 zones), and on sites of at least 1,000 square metres and with a maximum of 12 rooms.
- Parking requirements to be reviewed, with the aim of reducing the requirements, for boarding houses in Centre Investigation Areas in accessible locations (such as core areas and mixed housing areas).

²⁰ The R4 zone is not currently applied under any Northern Beaches LEPs. A revised LEP which integrates the existing LEPs may incorporate an R4 zone, as this is better reflective of existing housing character and types in some areas, and would allow for some higher density housing types in appropriate locations without the need to provide ground floor commercial uses (as is the case with the B4 zone currently).

Spatial approach for the location of boarding houses



Source: SGS, 2020.

32-34 quirk rd, Manly Vale



The Housing for Seniors or People with a Disability SEPP, refers to seniors as people aged 55 or over. Seniors housing is defined as residential accommodation that is (or is intended to be), used permanently for seniors. This policy also covers housing for people with a disability. The SEPP aims to provide all people in NSW, including older people and people with disabilities, with affordable and well-designed housing.

Locational controls

In addition to the zoning outlined in the table below, the SEPP contains additional location controls that limit where and how seniors housing can be developed, such as:

- requirement to be no more than 400 metres from services
- access pathways required to have an overall gradient of no more than 1:14
- public transport stop to be within 400 metres with a minimum standard of service per day
- additional features required in areas adjoining urban land
- minimum lot size of 1,000 square metres and a minimum site frontage of 20 metres
- maximum building heights.

Table 16:

Permissible zones under Northern Beaches LEPs for seniors housing under the SEPP

Manly LEP	Pittwater LEP	Warringah LEP
R1 General Residential	-	-
R2 Low Density Residential	R2 Low Density Residential	R2 Low Density Residential
R3 Medium Density Residential	R3 Medium Density Residential	R3 Medium Density Residential
-	R5 Large Lot Residential (adjacent to urban use zones)	-
B1 Neighbourhood Centre	B1 Neighbourhood Centre	B1 Neighbourhood Centre
B2 Local Centre	B2 Local Centre	B2 Local Centre
B3 Commercial Core	B3 Commercial Core	B3 Commercial Core
B4 Mixed Use	B4 Mixed Use	B4 Mixed Use
E4 Environmental Living	E4 Environmental Living	E4 Environmental Living
-	RU2 Rural Landscape (adjacent to urban use zones)	-
-	-	RU4 Primary Production Small Lots (adjacent to urban use zones)

Note: Seniors housing is currently permitted within the deferred lands area as identified within Warringah LEP 2000. Source: Northern Beaches LEPs, State Environmental Planning Policy (Housing for Seniors or People with a Disability) 2004.

Application in Northern Beaches LGA

Expected demand for seniors housing will require the supply of the right types of seniors housing development in the right locations.

Increasingly, seniors housing developments have taken the 'continuum of care' approach, allowing for residents to keep living in one place as they get older by providing different levels of care (from self-contained independent living units to nursing homes and palliative care) on one site. Brisbane City Council, for example, has amended their planning controls to encourage all-inclusive seniors housing development. We will encourage 'continuum of care' approach, especially for existing seniors housing sites with adequate land area and near centres. Development in inaccessible locations on the urban fringe will be discouraged; incentives will encourage this type of development near strategic and local centres with access to facilities and high-quality transport.

Case Study 1:

Retirement And Aged Care Incentives Scheme - Brisbane

Brisbane City Council is expecting a 50 per cent increase in demand for seniors housing by 2027. It has implemented initiatives to ensure adequate provision, prioritise best-practice design and encourage integrated facilities that include community facilities and other uses. These measures include:

- greater consideration of ageing in neighbourhood plans
- an incentive to reduce infrastructure charges levied on seniors housing
- a two storey height bonus for seniors housing in high-density locations.
- a two storey height bonus on a site 3,000 square metres or greater and a four storey height bonus for sites 7,000 square metres or greater in the Community Facilities zone, Specialised Centre zone or Emerging Community zone.



Proposed Ozcare development in Brisbane – 120 nursing beds and 142 retirement village units

References: Brisbane Times, 2019, 'Two towers for retirement and aged care planned for Newstead,' <u>www.brisbanetimes.com.au/national/queensland/two-towers-for-retirement-and-aged-care-planned-for-newstead-20190521-p51pkm.html</u>

Council will seek an exemption to some of the requirements under the SEPP and introduce local planning provision to encourage seniors accommodation. These provisions are:

- Seniors housing to be permissible in R2, B1, B2 zones only when they are within Centre Investigation Areas or within 400 metres from local centres on sites greater than 1,000 square metres.
- Seniors housing to be restricted in rural zones and environmental living zones, even when adjacent to urban use zones.
- Seniors housing to be restricted from R2, B1, B2 zones outside of a 800 metre radius of Centre Investigation Areas and 400 metre radius of local centres.
- Existing seniors housing in the R2 zone on sites that are larger than 3,000 square metres to be subject to additional height and density bonuses to incentivise redevelopment such as onsite nursing homes or upgrades to independent living units to include lifts.
- Allow additional height and density (maximum one additional storey) (subject to a more detailed urban design study) for seniors housing located in the R3, R4²¹ and B4 zones in Centre Investigation Areas. This could be completed as part of centre-specific planning work.
- Seniors housing and vulnerable facilities are to be avoided in high risk natural hazard areas or areas that have evacuation constraints.

²¹ As above, an integrated Northern Beaches LEP may incorporate an R4 zone, as this is better reflective of existing housing character and types in some areas, and would allow for some higher density housing types in appropriate locations without the need to provide ground floor commercial uses (as is the case with the B4 zone currently).



Source: SGS, 2020.

The Low Rise Housing Diversity Code aims to facilitate medium density housing. This targets the so-called 'missing middle' of housing, which is described as low-scale infill housing (such as dual occupancies, manor houses or multidwelling forms like terraces), and is focused on new lower scale housing, consistent with local character, in middle ring suburbs.

Under the code, typologies such as dual occupancies, manor houses and terraces are classed as complying development (meeting certain development standards) and can be fast-tracked in the R1 General Residential, R2 Low Density Residential, R3 Medium Density Residential, and RU5 Village zones.

Application in the Northern Beaches

We will seek an exemption to the Low Rise Housing Diversity Code and instead propose to meet demand for medium density housing through the strategies and implementation of this LHS through LEPs. These housing forms would still be permitted but targeted to appropriate locations around centres with good accessibility and amenity as identified through local planning.

Further detail on this approach is described in Section 7.

Figure 17: Medium density development typologies in the code



Source: NSW Government, 2017.

Multi-unit housing around Warriewood



Section 6: Sustainability Innovations



The change anticipated by new development can leverage greater sustainability and liveability outcomes, particularly given so many parts of the LGA will be subject to minimal change. Towards 2040 anticipates this by requiring sustainability innovations in planning proposals.

While extending the BASIX requirements for new development will reduce emissions and water use on a building by building basis, the Centres Renewal Framework allows us to consider sustainable development initiatives at the precinct level²². Future detailed planning work could consider initiatives in areas such as:

- using renewable energy and possibly decentralised energy sources
- reducing water use in general and encouraging water recycling
- integrating public open space into the planning of housing developments
- encouraging more active travel and less car use.
- embedding the circular economy in the design of buildings, infrastructure and the public domain to maximise the circularity of the materials used in construction.

Coordination with the NSW Government, agencies, institutions and regulators in the energy, water and transport sectors may be required. Typically, the regulatory environment rather than technology or the availability of alternative solutions can be a barrier to innovation.

Warriewood



Renewable energy

Council has an existing commitment for our community to have net zero emissions by 2050. Initiatives that could contribute to this target include investment in renewable energy systems at the precinct level, such as rooftop solar and shared energy storage. New development in centres such as Brookvale, Dee Why and Mona Vale could be the catalyst for such innovations.

Other councils in NSW and elsewhere have begun to invest in such programs.

Case Study 1: White Gum Valley Energy Initiatives - Fremantle, WA

White Gum Valley is a residential estate three kilometres south of Fremantle that is being developed as a sustainable demonstration precinct by the WA Government. The project aims to achieve net zero energy status, where the precinct generates as much energy as it uses.

Energy initiatives include:

- Energy efficient building design guidelines for detached dwellings, with a minimum 7-star NatHERS rating. Developers are required to demonstrate innovation in their proposals in both energy efficiency and providing renewable energy.
- Sustainability packages of \$10,000 to lots that comply with the design guidelines, covering the cost of upgraded photovoltaic (PV) systems and water management systems.
- Solar storage trial with three apartment buildings using battery storage technology and an equitable system to allow the sharing of energy by occupants. Residents can also monitor their usage in real time.
- Peer-to-peer trading trial, in partnership with the local network operator, where surplus energy can be traded across buildings to use locally.

Artists' impression of The White Gum Valley Estate

References: Low Carbon Living CRC, 2018, 'Guide to Low Carbon Precincts,' <u>www.lowcarbonlivingcrc.com.au/sites/all/files/</u> <u>publications_file_attachments/lclguide_precincts_web.pdf</u>

²² Application of the new Design and Place SEPP will also incorporate the use of BASIX.

Lismore City Council's Community Solar initiative developed the first joint council and community-owned solar farms in Australia, including the largest floating solar farm. Council established two 100 kilowatt farms – one as rooftop solar at the Goonellabah Sports and Aquatic Centre, and the other on the overflow ponds at the East Lismore Sewage Treatment Plant. The solar farms are a key part of Council's Renewable Energy Master Plan, which includes a target for Council to self-generate all of its electricity from renewable sources by 2023.

The solar farms were funded by 20 local investors under a unique financial model that sees the funds loaned to Council, to be repaid over seven years. The rooftop farm provides 15 per cent of the aquatic centre's power, and 12 per cent of the sewage plant's energy needs.



The largest floating solar farm in Australia

References: Low Carbon Living CRC, 2018, 'Guide to Low Carbon Precincts,' <u>www.lowcarbonlivingcrc.com.au/sites/all/files/</u> <u>publications_file_attachments/lclguide_precincts_web.pdf</u>

Water efficiency

Water sensitive urban design and passive and green roofs can support biodiversity and provide recreation opportunities while increasing water efficiency by retaining stormwater. Some cities overseas have introduced schemes that offer incentives for or require the provision of green infrastructure as part of new developments

Case Study 3:

City Of Toronto - Green Roof Bylaw

The City of Toronto's Green Roof By law uses financial incentives and penalties to ensure the uptake of green roofs. It stipulates specific Green Roof requirements, where new commercial, institutional and residential buildings with a gross floor area over 2,000 square metres require green roof coverage. Exemptions are only afforded if the developer pays \$200 a square metre charge for the area that is not covered by a green roof.

The approach is emblematic of a progressive green infrastructure policy that uses simple and direct guidelines coupled with financial penalties to ensure the integration of green roofs into the built environment.

Residential and non-residential (excluding industrial)

The green roof coverage requirement is graduated, depending on the size of the building. The table below shows how the requirement ranges from 20-60 percent of Available Roof Space for commercial, institutional and residential development.

Gross Floor Area* (Size of Building)	Coverage of Available Roof Space (Size of Green Roof)
2,000 - 4,999 m²	20%
5,000 - 9,999 m²	30%
10,000 - 14,999 m²	40%
15,000 - 19,999 m²	50%
20,000 m² or greater	60%

*Note: Residential buildings less than 6 storeys or 20 m in height are exempt from being required to have a green roo

References: City of Toronto, 2018, City of Toronto Green Roof Bylaw, 2018, <u>www.toronto.ca/city-government/planning-development/</u> official-plan-guidelines/green-roofs/green-roof-bylaw/ The Vermont housing estate is a greenfield development in Pitt Town in Sydney's north west. Pitt Town Water is an independent water utility that provides wastewater and water recycling services to the estate through a local recycled water network. Household wastewater and recycled water pipe systems are kept separate from drinking water, allowing for recycled water to be reused within the home rather than going down the drain. Wastewater and recycled water is also returned to a local water centre from where it can be used in the wider community, such as on gardens and sports fields.





Pitt Town Water Recycling System

References: Pitt Town Water, 2020, 'How it works,' www.pitttownwater.com.au/how-it-works-3/

Pulaski Park in Chicago has integrated water sensitive urban design. Bio swales that treat, retain and move stormwater from one place to the other through long narrow vegetated spaces can be used in precincts and parks to deliver ecosystem services. Introducing bioswale natural processes, such as natural water filtration, supports ecological processes by facilitating natural functions



Pulaski Park Bioswale

References: <u>www.asla.org/2018awards/455278-</u> <u>Re-Envisioning_Pulaski_Park.html</u> We already support the use of electric vehicles and the provision of charging stations around the LGA. Initiatives targeting sustainable mobility in other cities have also included restrictions and incentives to discourage the provision of parking spaces in new developments.

While car dependence is high in the Northern Beaches LGA, car ownership and use by younger people is declining, following the potential phenomenon of 'peak car' observed elsewhere in the world.²² Congestion is significant and, while public transport is constrained, it makes sense to push for less car use and more contained living, where access to a permanent vehicle is less important, to meet a growing market for car-less living. Requiring maximum parking rates, rather than minimums, or providing for modest contributions in lieu of parking for dedicated, offsite parking in areas away from a building can then decouple the demand for both parking and housing. For residents in new buildings with reduced or only visitor or car share parking, restrictions on onstreet parking permits would be necessary.

Case Study 6:

Nightingale development approved with no car parking

The Nightingale Model was developed by Nightingale Housing in Melbourne. Under this model, investors commit to financing an apartment project with capped profit margins, which are lower than the margins usually used by developers. Minimal marketing is used and buyers are engaged throughout the design and development process.

Apartments are designed to minimise ongoing maintenance and operational cost – and importantly do not have car parking. The first project developed under the model was approved to have no car parking as the site had excellent access to public transport. This encourages sustainable mobility, while also delivering housing that is environmentally and socially sustainable.



Nightingale Development approved with no car parking

References: www.nightingalehousing.org/nightingale-2

²³ Lee-Williams and Falconer, 2019, 'Peak car – what is it, and so what?' www.thefifthestate.com.au/urbanism/infrastructure/peak-car-what-is-it-and-so-what/ Citu Group is developing a Climate Innovation District in Leeds, which is aiming to be energy, water and waste efficient, while encouraging sustainable mobility and high amenity. The first release of housing has integrated green roofs, passive solar design and rain gardens, with 100 per cent renewal household energy. Other design initiatives mean that household energy and water consumption can be better monitored and controlled.

The district's location means that most residents will be able to walk or cycle to work. However, as an additional incentive to decrease car usage, to obtain a parking space residents are required to pay a fee of £15,000. Charging infrastructure will also be provided to support electric vehicles.



Leeds Climate Innovation District Development by Citu

Urban renewal creates opportunities to bring about improvements to public areas as new buildings are developed. This can include water sensitive urban design to support natural ecological processes, programs to foster urban greening, and master planning at the precinct level to integrate local environmental and social benefits.

Case Study 8:

Kensington and Kingsford town centres strategy

The Kensington and Kingsford Town Centres Strategy shaped the transition of Kensington and Kingsford to take advantage of better access and public domain improvements created by the South-East Light Rail.

An analysis of the public domain identified opportunities for 'urban repair' with the introduction of rear lanes, mid-block parks, and new street extensions to encourage street activity, walking and cycling around apartment buildings as part of a wider network of green infrastructure.



Public domain and open space analysis for Kensington to Kingsford

References: Hill Thalis, JBA, Bennett and Trimble, Kensington to Kingsford

Design Competition, see <u>www.hillthalis.com.au/projects/k2k</u>

NSW Government Architect, 2017, 'Greener Places,' <u>www.governmentarchitect.nsw.gov.au/resources/</u> ga/media/files/ga/discussion-papers/greener-places-discussion-draft-2017-11.pdf

Financial implications

Financial implications associated with these types of initiatives could be costs to Council and/or costs to developers who may be required to incorporate additional features.

The implications of additional costs and their impact on the viability of development will need to be considered during detailed planning phases. We will aim to establish a platform for innovation and seek external funding and assistance to achieve ambitious sustainability outcomes.

Ocean Street, Narrabeen



Section 7: Future Planning



The approach to planning for new housing on the Northern Beaches will focus on:

- concentrated medium and higher density urban renewal around accessible centres along the B-Line corridor
- opportunities for low to medium density housing around local centres within a 400m radius, subject to environmental and other constraints
- Locating boarding houses and seniors housing around centres and in accessible areas, subject to conditions.

Spatial approach

The spatial approach to planning for new housing across the LGA is based on:

- the Centre Investigation Areas
- low-rise housing diversity areas within 400m of local centres
- R2 Low Density Residential zones outside of centres
- E4 Environmental Living and R5 Large Lot Residential zones.

Planning for new housing under this framework will require a staged approach, focusing on selected centres over time.

Allocation of typologies

Different housing typologies are promoted in these four areas across the LGA.

Low density typologies, such as single and secondary dwellings, will be limited to existing low density and environmentally sensitive residential areas – E4 Environmental Living and R5 Large Lot Residential and R2 Low Density Residential zones outside of centres.

Low-medium density typologies, such as attached dual occupancies, will be investigated within 400m of local centres (excluding absolute constraints), and in Influence Area precincts within Centre Investigation Areas (to be defined through future planning) around the B-Line centres.

Medium density housing types, including those targeted by the Low Rise Housing Diversity Code, will be concentrated in influence areas and mixed housing areas around the B-Line centres (to be defined through future planning).

Higher density **residential flat buildings** will be focused on Mixed Housing and Centre Core precincts (to be defined through future planning) around the B-Line centres.

Special (SEPP) housing types

Seniors accommodation will be limited to areas within 400m of local centres and in Centre Investigation Areas, with some facilities potentially permitted on R2-zoned sites subject to certain conditions.

Boarding houses will be limited to Centre Investigation Areas, and within 400 metres of local centres if delivered by social and community housing providers subject to certain conditions.

The matrix included overleaf illustrates the spectrum of dwelling types and where new housing will be located. This compares the approach of the LHS to the typologies identified in the Low Rise Housing Diversity Code to illustrate how these types of dwellings will be provided under the LHS.



Freshwater

Avalon, Newport, Warriewood and Belrose - low-rise housing diversity areas: Dual occupancies, seniors and boarding houses within 400m of centres

Mona Vale - Centre Investigation Area: No increase to building heights proposed Detailed planning to investigate medium density around centre

Narrabeen - Centre Investigation Area:

Detailed planning to investigate medium density around centre

Dee Why - Centre Investigation Area:

No further high density proposed Detailed planning to investigate medium density around centre

Brookvale - Centre Investigation Area:

Detailed planning to investigate higher density housing in centre and medium density around centre

Detailed planning to investigate medium density around centre

Balgowlah - low-rise housing diversity area: Support existing housing diversity in 400m around centre

Manly - low-rise housing diversity area: Support existing housing diversity Place Plan to identify opportunities to balance visitor

Frenchs Forest - NSW Government-led precinct:

Separate precinct planning process to investigate higher density housing in centre

Forestville and Beacon Hill - Future Centre **Investigation Areas:**

Detailed planning to investigate centre renewal and medium density subject to new B-Line



Source: SGS, 2020.

Low Density

		Single dwelling	Two dwellings (secondary dwellings)	Dual occupancy (attached, and detached on corner lots only)	Multi-dwelling housing and terraces
	E4 Environmental Living/R5 Large Lot Residential				
	Rest of R2 zones (not including Centres investigation areas and not within 400m of centres)				
	Low rise housing diversity areas (400m radius)				
eas	Throughout Centre Instigation Area (approx. 800m radius) (Influence area)				
Cantre investigation areas	Mixed housing - 400-800m (to be defined through detailed planning work)				
Can	Higher density and mixed use core - up to 800m (to be defined through detailed planning work)		1 		
			(·	A









Single dwelling

Two dwellings (secondary dwellings) Dual occupancy (detached/side by side)

Multi-dwelling housing and terraces

Source: SGS, 2020.



Residential flat

buildings



Dual occs

(one above other) Manor house

Centre investigation areas include land that is:

- within 800-metre radius of a centre
- free of natural hazards or constraints such as ecological, hydrological, geological and bushfire constraints
- of a topography suited to intensification.

Council and NSW Government data has informed absolute and potential risks, which in turn informed the Centre Investigation Area boundaries.

Significant constraints

Areas subject to significant constraints include high risk flooding areas, bushfire-prone land, land with Class 1 acid sulphate soils, areas with existing vegetation and wetlands, and coastal risk areas. These are not suitable for development.

Manageable constraints

Manageable constraints may constrain but do not absolutely preclude development - risks can be managed to allow for new housing. This includes areas subject to low to medium flooding risk, land that contains Class 2 to Class 5 acid sulphate soils, coastal environmental areas or land identified as having landslip risks.

Pittwater Road, Dee Why


Centre investigation area boundaries

Different development precincts (centre core, mixed housing and influence areas) in centre investigation areas are subject to different opportunities and constraints, including topography, walkable distances, heritage items, strata properties, open space and lot size.

1. Centre core

Centre cores can include larger sites suitable for major redevelopment, or areas that have potential for site amalgamations. Targeting new centre core development in areas that already have existing affordable housing should be avoided.

Centre cores are suited to mixed use and the more intensive housing, such as apartments.

Figure 19:

Centre Core Typologies



Source: Architectus, 2019.

Detailed planning for each of the Centre Investigation Areas will be undertaken to determine the appropriate level of new development that can be accommodated.

The Centre Investigation Areas will be considered individually to reflect their unique character and circumstances. This means that what is planned in lowerdensity locations such a Mona Vale will not be the same typologies or scale of development as larger and higher density centres like Dee Why or Brookvale.

2. Mixed housing

Mixed housing is suited to relatively larger lot sizes, typically sites of over 600 square metres, and ready for redevelopment without the need for site amalgamations.

Housing typologies in should include intermediate and medium density housing forms, such as townhouses and small scale apartments of two to four storeys.

Figure 20:

Mixed Housing Precinct Typologies



Source: Architectus, 2019.

3. Influence area

Influence area precincts should accommodate compact infill housing suited to the character of existing detached housing areas, such as dual occupancies, terraces, semi-detached dwellings, and manor homes. Ideally, these areas are Torrens titled.

Figure 21:

Influence Area Typologies



Source: Architectus, 2019

Applying the Framework

The Centres Renewal Framework will be applied to each centre via a structure plan and/or place planning process to optimise each centre's unique characteristics.

The process of identifying the centre boundary is illustrated for each centre.

The sequence of detailed future planning may change, but is likely to begin with Brookvale, Dee Why and Mona Vale. Brookvale may be the first centre to be investigated, to complement and integrate with the Brookvale Structure Plan with the Brookvale Structure Plan and the Draft Northern Beaches Employment Strategy.

Brookvale

Brookvale is subject to some flooding risks, and its west is constrained by native vegetation. There are also some coastal inundation risks to the south.

Figure 22: Brookvale Coastal and Ecological Constraints



Aligned with the native vegetation, there are bushfire and landslip risks to the west, as well as acid sulphate soils and landslip risks in some other areas. The constraints exclude a lot of the western side of the area from consideration.

Figure 23: Brookvale Geological and Bushfire Risks



There are around 2,030 dwellings within the wider renewal area, with remaining capacity for around 740 dwellings under current planning controls.

Employment land must be retained in Brookvale. Separate studies will be considered in planning for Brookvale's future so that any new housing complements Brookvale as the LGA's main employment centre. Traffic congestion is another issue, as is the need for open space and community facilities.

Figure 24: Brookvale Development Investigation Area



Source: SGS, 2020

Dee Why

Dee Why includes some areas of flooding risk, coastal environment areas, and some patches of existing vegetation.

Figure 25: Dee Why Coastal and Ecological Constraints



There are some acid sulphate soil areas on the periphery of the 800 metre radius and some sloping areas which have a risk of landslips.

Figure 26:

Dee Why Geological And Bushfire Risks



This excludes some areas to the north-east and patches to the south from consideration. Across this broad area there is around 10,040 dwellings, with capacity for another 1,340 under current planning controls.

Planning in Dee Why must retain and restore its character given the substantial development in recent years. Its proximity and relationship to other centres, including the corridor between Dee Why and Brookvale, and smaller centre of Narraweena to the west, should be optimised and congestion managed. Other constraints and issues include:

- the high number of walk-up flats to the east of Pittwater Road, unlikely to redevelop and already providing good quality and relatively affordable medium density housing in an attractive location
- steep inclines to the west of the centre that may not encourage people to walk
- minimising conflicts between employment uses along Pittwater Road and residential development.

Figure 27:

Dee Why Development Investigation Area



Mona Vale

Mona Vale has some low level flooding risks at its core and more constraints, including coastal and higher risk flooding, at its edge. It offers large open space areas.

Figure 28: Mona Vale Coastal And Ecological Constraints



There are acid sulphate soils across the area, though mostly not in the highest risk category.

Figure 29: Mona Vale Geological And Bushfire Risks



The above constraints exclude some areas to the south east from consideration. The area contains around 3,240 dwellings, with capacity for approximately 880 under the existing planning controls. Building height increases are not proposed for Mona Vale. Mona Vale is subject to traffic congestion, given existing pressure points around schools and intersections on Pittwater Road. Retaining employment uses around Mona Vale is essential as it is the major centre for the northern part of the LGA. Appropriate infrastructure to support new housing development should be considered.

Figure 30: Mona Vale Development Investigation Area



Source: SGS, 2020

Manly Vale

Manly Vale is surrounded by flooding and coastal risks and is constrained in the west by vegetation.

Figure 31:

Manly Vale Coastal And Ecological Constraints



Similarly, there are bushfire risks to the west, and some landslip and acid sulphate risks elsewhere.

Figure 32: Manly Vale Geological And Bushfire Risks



Source: SGS, 2020

Significant constraints exclude several areas. Within this broad area there are around 3,640 existing dwellings, with capacity for a further 220 under existing planning controls. Future planning for Manly Vale should consider relationships and connections with nearby Balgowlah, which is not served by the B-Line. Other issues include congestion and the potential additional impact that could be generated from the Western Harbour Tunnel and Beaches Link. The location of supermarkets alongside big box retail and other commercial uses around the core generates traffic, and will need to be considered, as will the potential for supporting community and other infrastructure.

Figure 33: Manly Vale Development Investigation Area



Narrabeen

Narrabeen is constrained by coastal and flooding risks.

Figure 34: Narrabeen Coastal And Ecological Constraints



There are some areas of acid sulphate soils within the 800 metre radius.

Figure 35: Narrabeen Geological And Bushfire Risks



These risks exclude a lot of the land in the area from consideration, though there may still be some opportunities for renewal. Within the identified boundary there are around 1,140 existing dwellings, with capacity for

a further 210 under existing controls.

The key issue for the future planning for Narrabeen is its coastal risks, particularly over the longer term with climate change and associated sea level rise. Adding to these constraints is traffic congestion and the high number of strata properties, particularly facing the beach, which may limit the ability for sites to be redeveloped due to financial feasibility. This would need to be investigated in future planning.

Figure 36: Narrabeen Development Investigation Area



Source: SGS, 2020

Potential future housing densities and capacities

A high level assessment of potential future dwelling densities and capacities considered the capacity analysis across the LGA and existing dwelling densities of unconstrained areas.

Potential future densities draw from increased density assumptions for different areas and a comparison with similar areas. An ambitious scenario and then slightly more conservative scenario were tested. Different development take up rates were assumed for centre cores, mixed housing and influence area to reflect that higher density precincts are more likely to turn over than lower density precincts. These assumptions will need to be tested and confirmed during future detailed design work in each centre, as they can impact the calculation of potential dwelling uplift.

The high level assessment suggests that applying the Centres Renewal Framework could increase housing capacity in the range of 1,550 to 2,930 dwellings. Though confirmation is subject to detailed planning, this range would allow Council to address the identified capacity gap of 275 while helping to meet projected future housing demand to 2036.

Table 17:

Potential future housing densities around centre for more detailed investigation

Centre	Current dwelling density across investigation area (excl. absolute risks)	Existing maximum dwelling density (under current controls)	Potential future density across investigation area (dwellings per hectare)	Potential additional dwelling capacity generated
Brookvale	17	23	30 - 33	840 - 1,240
Dee Why	70	80	81-84	260 - 570
Mona Vale	25	32	33 – 35	180 - 400
Manly Vale	39	41	44 - 45	240 - 440
Narrabeen	77	92	94 - 111	30 – 280
Total additional cap	acity			1,550 - 2,930

Source: SGS, 2020.

Tailored application

Each centre would see different levels of development and housing typologies, reflective of their character and role. Brookvale may be able to accommodate some more high density residential and mixed use development. Dee Why and Mona Vale may see some more medium density development, whereas smaller centres like Narrabeen (or Forestville and Beacon Hill into the future) would only accommodate low rise, low to medium density development in mixed housing and influence areas. While it is currently a smaller centre, Manly Vale could also be considered for some medium density development given its accessibility to the B-Line and its role as a future centre in the LGA. The assessment of potential future densities and yields is undertaken at a high level and requires more detailed planning that applies the Centres Renewal Framework based on detailed built form and design analysis to identify core, mixed housing or influence areas.

This work must consider and respect existing local housing character and identify opportunities for a mix of housing, including new approaches to elements such as the public domain and parking provision. Other elements that will need to be explored include:

- opportunities for co-design with the community, allowing for meaningful engagement and input
- broader Towards 2040 targets, including in emissions reduction, reducing water use, increasing use of public and active travel, reducing space for car parking, provision of open space and overall liveability
- potential to engage with WaterNSW, Transport for NSW, and energy and other utility providers as partners in innovation
- how to generate a liveability dividend, and associated local funding plans
- determining affordable housing contributions or the potential to provide affordable housing through voluntary planning agreements
- funding and implementation innovations, such as achieving other community benefits through housing uplift, or pooled development approaches.

Timing

In the short to medium term, the focus will be on Brookvale, Dee Why, Mona Vale, Manly Vale and Narrabeen – ideally focusing on the strategic centres first. If a second B-Line route is confirmed, additional detailed investigations for Forestville and Beacon Hill could also be undertaken. Ocean Street, Narrabeen



Section 8: Strategy and Implementation



The vision for housing on the Northern Beaches – summarising the principles from Towards 2040 - is to:

- meet the growing and changing housing needs and preferences of existing and new residents
- provide additional housing within the existing urban footprint and respect local heritage, environmental features and built form
- provide most new homes in and near centres with good access to public transport and where walking and cycling to shops and services is easy
- provide a mix of housing to enable young and older residents to stay in their communities on the Northern Beaches
- provide more affordable housing
- help to achieve lower energy and water use, less car dependency and new and enhanced open space.

The Strategy includes five priorities:

- Priority 1: Housing target meet District Plan and 6-10 year housing target.
- Priority 2: Detailed planning for centres - establish capacity to accommodate housing demand in centres
- Priority 3: Social and affordable housing - encourage more affordable housing and allow for boarding houses in appropriate and accessible locations.
- Priority 4: Precinct sustainability and housing - investigate and support sustainable housing precincts and maintain the environmental qualities and other features of the Northern Beaches.
- Priority 5: Seniors housing incentivise seniors housing in accessible locations.

Priority 1 Housing target

The 6-10 year housing target is based on the projected future demand for housing in the LGA derived from population projections. The target can be achieved within the current planning framework. It will be updated with the release of the next population projections, which will provide an indication of the short-term impact of COVID-19 and reduced migration.

A 20-year projection of future housing demand has also been identified.

The GSC has also set a 6-10 year target for the Northern Beaches LGA of 3,500-4,000 dwellings.

Table 18:

Housing target for Northern Beaches LGA

Timeframe	Gross target	Supply 2016-20*	Net target
2016-21 (District Plan)	3,400	2,835	565
2021-26 target** - from new DPIE projections	3,582	-	3,582
2026-36 projection of housing demand (based on DPIE projections)	8,949	-	8,949

Source: SGS, 2020.

Table 19:

Actions - Reporting on housing target

Action	Indicative timing	Dependencies and responsibilities
Deliver the District Plan (3,400 dwellings) and 6-10 Year (3,582 dwellings) housing target ²⁴	2021, 2026	Council
Report annually on development to track yields and housing mix - by centre, LGA-wide, and for boarding houses and seniors living.	Annually	Council

Source: SGS, 2020.

 * $\,$ Based on DPIE Housing Monitor dwellings completions and Council data.

Includes secondary dwellings, boarding houses and seniors living dwellings.

** Provisional 6-10 year target, to be updated following release of new DPIE population projections that show the short term impact of COVID-19 and reduced migration on Sydney's overall population growth.

²⁴ The target for 2016-21 is included in the North District Plan, while the 2021-26 housing target is based on the projected future demand for housing in the LGA derived from DPIE's population projections.

Priority 2 Detailed planning for centres

Identifying opportunities for urban renewal will account for the unique circumstances, characteristics and constraints for housing in each relevant centre. This process will require collaboration with the community and NSW Government agencies.

This detailed planning will be based on the **Centres Renewal Framework**. It **must facilitate high quality design and supporting infrastructure** such as open space, community facilities and transport, and how this can be funded and provided. This may require statutory planning changes to mandate design standards or the provision of certain facilities alongside new development in certain areas, as well as further engagement with NSW Government agencies to plan and facilitate larger infrastructure upgrades. This work could be undertaken initially in one centre in the next year – likely Brookvale to align with the completion of the Brookvale Structure Plan. Proposed timing for the next stages of planning are noted below.

As each of these investigations is completed, LEP controls will be updated and number of new dwellings reported.

This action includes the proposed exemption from the Low Rise Housing Diversity Code.

For some low-rise housing diversity areas, Council may explore implementing these within the new LEP scheduled for commencement mid-2023. This would mean new dual occupancies, seniors housing and boarding houses could contribute to the 6-10 year housing target.

Table 20:

Actions - Detailed planning for centres

Action	Indicative timing	Dependencies and responsibilities	
Complete structure plan and update LEP for Brookvale	2021-25		
Undertake planning analysis and update LEP controls for Dee Why.	2025-36	Council, DPIE, Transport for NSW,	
Undertake planning analysis and update LEP controls for Mona Vale.	2025-36	NSW OEH, Water NSW,	
Undertake planning analysis and update LEP controls for Manly Vale.	2025-36	Essential Energy, Infrastructure NSW	
Undertake planning analysis and update LEP controls for Narrabeen.	2025-36		
Report against capacity gap.	Ongoing as each investigation/LEP update is completed	Council	
Use LHS to inform review of development contributions plans to fund infrastructure upgrades from new development	Ongoing	Council, DPIE	
Report annually on development by centre to track yields and housing mix.	Annually	Council	
Seek exemption from the Low Rise Housing Diversity Code and implement proposed LHS alternatives	2021-22	Council, DPIE	
Comprehensive LEP to implement low rise housing diversity areas where appropriate	2023	Council	

Priority 3 Social and affordable housing

A comprehensive strategy

Formally adopting a social and affordable housing target that matches the 1,880 projected additional dwellings needed by 2036 (or around 95 dwellings per year) is one way to address the shortfall in social and affordable housing.

We will also implement Council's endorsed target of 10 per cent of dwellings in new planning proposals to be affordable rental housing.

In addition to affordable housing contributions we will investigate:

- contributions negotiated through planning agreements
- wider adoption of inclusionary zoning
- redeveloping underutilised or surplus Council-owned sites or community-owned sites (not including open or green space)
- direct cash contributions
- joint ventures with community or private partners.

Table 21:

Actions - Social and affordable housing

Boarding houses

Boarding houses, with the right controls in place, can provide an affordable housing option. The following proposed planning provisions could facilitate the supply of boarding houses:

- allow boarding houses only in locations close to public transport and services, recognising that new generation boarding houses can provide a smaller unit option, and look to reduce parking requirements to minimise traffic impacts and reduce the cost of construction
- create opportunities for traditional boarding houses for low income households in the R2 zone only when they are within a kilometre of a local centre and when they are developed by a social or community housing provider.

This broad approach is also in line with what is being proposed through the Housing Diversity SEPP.

Action	Indicative timing	Dependencies and responsibilities
Adopt social and affordable housing target (e.g. minimum 1,884 new dwellings between 2016 and 2036).	2021	Council
Prepare and implement affordable housing contribution scheme.	Ongoing	
Develop comprehensive Affordable Housing Action Plan, which considers additional measures to help achieve SAH target.	2021-22	Council, CHPs, LAHC
Seek exemption from the State Environmental Planning Policy (Affordable Rental Housing) 2009 regarding the provision of boarding houses.	Ongoing	Council, DPIE
Amend LEP to provide local controls for the provision of boarding houses.	After exemption is granted	Council, DPIE
Comprehensive LEP to implement low rise housing diversity areas including boarding houses where appropriate	2023	Council
Review and seek lower parking requirements for boarding houses for R3, R4 ²⁵ and B4 sites in centre investigation areas.	After the exemption is granted	Council, DPIE

²⁵ As above, an integrated Northern Beaches LEP may incorporate an R4 zone.

Priority 4 Precinct sustainability and housing

Targets to improve and maintain the sustainability and liveability of the Northern Beaches can be achieved if we leverage the opportunities that new housing development provides.

We will investigate and understand how different approaches and initiatives may be able to be incorporated into urban renewal opportunities in the LGA. This will include a framework for development proponents to consider and work within, prepared in parallel with and as an input to the first detailed centres planning exercise.

LEPs and DCPs will reflect these targets and be utilised to enforce sustainability requirements.

We will undertake hazard studies and develop a Resilience Plan for the Northern Beaches. These studies may identify that some centres are unsuitable for further development (such as Narrabeen which is heavily constrained).

Table 22:

Actions - Sustainability

Action	Indicative timing	Dependencies and responsibilities	
Develop a framework to integrate sustainable development and new housing at a precinct level to inform detailed centres planning (linked to Priority 2 actions), including updating LEP/DCP controls.	In conjunction with Priority 2	Council, DPIE, Transport for NSW, NSW OEH, Water NSW, Essential Energy, Infrastructure NSW	
Establish baseline sustainability metrics (e.g. energy, water, waste, car use, walking etc) for each centre as part of detailed centres planning and report on achievements and performance.	In conjunction with Priority 2		
Update LGA wide controls through LEPs/DCPs to achieve sustainable development outcomes consistent with Towards 2040, including consideration of financing sustainability initiatives through developer contributions.	In conjunction with Northern Beaches DCP/LEP preparation	Council, DPIE	
Undertake hazard studies and develop a Resilience Plan for the Northern Beaches.	In conjunction with Northern Beaches DCP/LEP preparation	Council	

Priority 5 Seniors housing

Planning should encourage seniors housing in accessible locations, enable the industry trend of the 'continuum of care' approach on larger sites, and provide incentives for seniors housing that meets the needs of the community. A Northern Beaches-specific approach will:

- allow for the redevelopment and upgrade of older and larger seniors housing sites to increase the supply of all-inclusive seniors housing including residential care facilities
- incentivise the provision of seniors housing in accessible locations in Centre investigation Areas, within multi-level developments
- limit developments in inaccessible locations that will not meet the need of older people, including avoiding areas of high risk natural hazards or with evacuation constraints.

The approach will be reflected in LEPs and DCPs.

Table 23:

Actions - Planning for seniors housing

Action	Indicative timing	Dependencies and responsibilities
Seek exemption from State Environmental Planning Policy (Housing for Seniors or People with a Disability) 2004 regarding the provision of seniors housing.	Ongoing	Council, DPIE
Amend LEP to provide local controls for the provision of seniors housing.	After the exemption is granted	Council, DPIE
Comprehensive LEP to implement low rise housing diversity areas with seniors housing where appropriate	2023	Council

Potential statutory implementation pathways

Planning controls and development standards are encoded in LEPs, SEPPs and DCPs. These will need to be amended to implement the LHS. This process will be determined through further study; potential pathways are noted on the following pages.

Staged approach to implementation

Implementation will occur in several stages:

1. More detailed investigations and masterplans for centres to identify precincts within

centre investigation area boundaries and any centre-specific design controls.

- 2. LEP, SEPP and DCP amendment to create a framework consistent with this land use planning approach, including permissible development in each zone.
- **3.** Staged LEP and DCP amendments to implement masterplans for each centre.

Potential land use zones, dwelling permissibility and other amendments are noted in the next table.

Table 24:

A potential framework for statutory implementation of the LHS approach in LEPs

Area	Current land zone	Suggested land zone	Permitted residential land uses	Additional amendments needed	Timing for implementation
Environmental Living/ Large Lot Residential	E4 or R5	E4 or R5	Single dwellings and secondary dwellings		ТВС
Rest of R2 zones (not including Centre Investigation Areas and not within 400m of centres)	R2	R2	Single dwellings and secondary dwellings	An additional LEP clause will be needed to restrict dual occupancy	TBC
Within 400m of local centres (Residential areas)	R2	R2	Single dwellings, secondary dwellings and dual occupancies	development in the R2 zone to the areas around local centres.	ТВС
Within 400m of local centres (Centres outside of centre investigation areas)	B1 or B2	B1 or B2	Lower density shop-top housing	The scale of development will be restricted through building height and FSR controls	TBC
Influence area	Mostly R2	R3	Dual occupancies, multi dwelling housing, manor houses, attached dwellings	An additional LEP clause will be needed to restrict dual occupancy development in the R2 zone to the 1kn area around investigation areas.	Later amendments n
Mixed housing	Mostly R2 or R3	R1	Multi dwelling housing, manor houses, lower density residential flat buildings	The scale of development to be restricted through height of building and FSR controls. Development outcomes to be tailored through detailed DCP controls.	Later amendments
Centre Core	Varies	R4 ²⁶ or B4	Higher density residential flat buildings and shop-top housing		Later amendments

²⁶ As noted above, the planned consolidated LEP for the LGA may need to include an R4 zone to better represent housing character in the Northern Beaches, and allow for some higher density housing forms in appropriate locations without the need to provide ground floor commercial space (as is the case with the current B4 zone).

SEPP amendments

Amendments to SEPPs will also be needed to implement this housing strategy. Potential pathways to achieve this are listed in the table below.

SEPPs are created by the NSW Government. Councils can amend SEPPs through planning proposals, which must be approved by the Planning Minister. Pathways to achieve this are noted in the next table. The core planning proposal to implement this strategy's land use planning framework will include LEP amendments as well as SEPP amendments applying only to the Northern Beaches LGA. These amendments will need to be discussed with DPIE.

Table 25:

Potential SEPP amendments

SEPP	Reason for amendment	Potential amendment mechanism
State Environmental Planning Policy (Affordable Rental Housing) 2009	To restrict boarding houses to appropriate locations – higher density zones and near identified centres.	Additional clause in the SEPP applying only to the Northern Beaches restricting permissibility of Boarding Houses using a set of spatial criteria.
State Environmental Planning Policy (Housing for Seniors or People with a Disability) 2004	To restrict seniors housing from occurring in inaccessible locations on the urban fringe, instead encouraging it in highly accessible areas.	Additional clause in the SEPP applying only to the Northern Beaches restricting permissibility of seniors housing using a set of spatial and property size criteria.
State Environmental Planning Policy (Exempt and Complying Development Codes) 2008	To exempt the Northern Beaches LGA from the application of the Low Rise Housing Diversity Code, as capacity for new low rise medium density housing is being created through local planning controls.	Additional schedule to the SEPP specifying that low rise medium density housing in the Northern Beaches cannot be approved through complying development.

Appendix 1 Proposed Housing Diversity SEPP

The proposed Housing Diversity SEPP will provide definitions and planning provisions for build-to-rent, purpose-built student housing and co-living.

Affordable rental housing provisions

The following changes are proposed to the Affordable Rental Housing SEPP:

- All boarding houses need to be affordable and managed by a registered not-forprofit community housing provider.
- The requirement for boarding houses to be mandated in the R2 Low Density Residential zone will be removed (LAHC will still be able to develop in the R2 zone, regardless of whether an LEP allows or prohibits boarding houses in that zone).
- The (FSR) bonus for new boarding houses to a maximum 20 per cent will be removed.
- There will be a simplified process to allow for an existing dwelling to be used as a group home.
- The provisions of Part 3 will be updated so that councils can mitigate the loss of affordable housing by levying monetary contributions.
- Councils will be able to set the maximum size for a secondary dwelling in a rural zone.

Seniors housing provisions

Prior to the introduction of the proposed Housing Diversity SEPP, two recent changes were made to the Seniors SEPP:

- New proposals for seniors housing on land within the Metropolitan Rural Area (MRA) are prohibited.
- The Heritage Conservation Area moratorium (the Seniors Housing SEPP did not apply in Heritage Conservation Areas in Greater Sydney) is extended to 1 July 2021.

The following changes are proposed to the Seniors SEPP (under the Housing Diversity SEPP):

- update the provisions of Schedule 1 -Environmentally Sensitive Land to align with current legislative and planning conditions
- extend the validity of a site compatibility certificate (SCC) to five years, provided a development application is lodged within 12 months of the date on which the SCC is issued
- clarify how the SEPP applies to land being used for the purposes of a registered club
- amend the 'location and access to facilities' provisions so that point-to-point transport such as taxis, hire cars and ride share services cannot be used for the purpose of meeting the accessibility requirements
- clarify that development standards in a local environmental plan prevail to the extent of any inconsistency with the SEPP.

Appendix 2 Options Considered

Concentrated urban renewal in and around strategic centres

- Housing growth and a diversified dwelling mix concentrated around Dee Why, Brookvale, Manly, Mona Vale, and Frenchs Forest
- Additional housing for people to live near jobs, services and transport.
- Higher densities could be included in mixed use zones and around centre cores, and medium density within walking catchment.
- In Mona Vale, housing must respect the village character of the centre.

Dispersed urban renewal in and around all centres

- Increased medium density development around centres.
- Future housing in Manly Vale and Forestville with access to current and proposed B-Line routes. Height limits of four to six storeys around centre core, and medium density within walkable catchment.
- Avalon, Newport, Narrabeen, Balgowlah and Freshwater could see medium density integrated with existing character.
- Opportunities for more medium density and diversity around Terrey Hills, Belrose and Beacon Hill could be investigated, subject to environmental constraints.
- This option has a focus on increasing housing diversity and choice in more locations.

Transport-centric renewal

- Renewal focused around current and future public transport stops and corridors.
- Opportunities for additional housing could be considered around:
- potential B-Line stops (such as Beacon Hill) and current B-Line stops (such as Warriewood)
- centres identified in the dispersed urban renewal option (such as Manly Vale and Forestville).

'Missing middle' housing diversity

- Allows for attached and detached dual occupancies in residential low density zones across the LGA, with a minimum lot size of 450sqm.
- Some areas could see attached and multiunit housing as infill development on sites with a minimum lot size of 600sqm
- Environmental living areas could be applied to areas where infill development should be limited due to environment/character, and development standards to mandate minimum deep soil zones and vegetation.
- This option would work well with the dispersed urban renewal option that would also promote medium density housing around local centres.









Appendix 3

Summary of potential uplift by centre

Table 26:

Summary of potential uplift by centre

Centre	Approx. land area (ha) of investigation area	Existing number of dwellings	Estimated remaining capacity under current controls	Potential additional capacity range applying Centres Renewal Framework
Brookvale	122	2,033	739	840 – 1,240
Dee Why	143	10,035	1,338	260 – 570
Mona Vale	129	3,240	879	180 - 400
Manly Vale	94	3,635	217	240 - 440
Narrabeen	15	1,140	210	30 – 280
Total	503	20,083	3,383	1,550 – 2,930

Warriewood







northern beaches council

northernbeaches.nsw.gov.au